REVIEW ARTICLE



Inverse problems in blood flow modeling: A review

David Nolte^{1,2,3}

| Cristóbal Bertoglio¹ [©]

¹Bernoulli Institute, University of Groningen, Groningen, The Netherlands

²Center for Mathematical Modeling, Universidad de Chile, Santiago, Chile ³Department of Fluid Dynamics, Technische Universität Berlin, Berlin, Germany

Correspondence

Cristóbal Bertoglio, Bernoulli Institute, University of Groningen, Groningen, The Netherlands.

Email: c.a.bertoglio@rug.nl

Funding information

European Commission; Horizon 2020; European Union; European Research Council

Abstract

Mathematical and computational modeling of the cardiovascular system is increasingly providing non-invasive alternatives to traditional invasive clinical procedures. Moreover, it has the potential for generating additional diagnostic markers. In blood flow computations, the personalization of spatially distributed (i.e., 3D) models is a key step which relies on the formulation and numerical solution of inverse problems using clinical data, typically medical images for measuring both anatomy and function of the vasculature. In the last years, the development and application of inverse methods has rapidly expanded most likely due to the increased availability of data in clinical centers and the growing interest of modelers and clinicians in collaborating. Therefore, this work aims to provide a wide and comparative overview of literature within the last decade. We review the current state of the art of inverse problems in blood flows, focusing on studies considering fully dimensional fluid and fluid-solid models. The relevant physical models and hemodynamic measurement techniques are introduced, followed by a survey of mathematical data assimilation approaches used to solve different kinds of inverse problems, namely state and parameter estimation. An exhaustive discussion of the literature of the last decade is presented, structured by types of problems, models and available data.

KEYWORDS

blood flows, inverse problems, mathematical modeling, medical imaging

1 | INTRODUCTION

Cardiovascular disease (CVD) is the major cause of death globally.¹ Alone in Europe, CVD causes 3.9 million deaths pear year, accounting for 45% of deaths from all causes. The estimated overall cost of CVD for the economy of the European Union is \notin 210 billion.²

CVDs generally alter the blood circulation by redirection or obstruction of the blood flow due to malformations of the heart, vessels or heart valves, by modification of the tissue properties (e.g., stiffness, lesions) or due to building up of arterial plaques. Accordingly, the diagnosis of CVD assesses hemodynamic properties of the diseased vessels. For instance, valvular stenosis, narrowing of vessels due to congenital abnormalities or atherosclerosis can cause oscillatory flow disturbances and turbulence leading to an *increase in hemodynamic pressure gradient* and thus an increased

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

© 2022 The Authors. International Journal for Numerical Methods in Biomedical Engineering published by John Wiley & Sons Ltd.

^{2 of 42} WILEY-

cardiac load. Therefore, the pressure gradient is extensively used as a diagnostic indicator of the severity of blood flow obstructions.^{3–7}

Another example is the assessment of the *arterial wall stiffness* (AWS), which is an extremely important indicator for the early diagnosis of hypertension,⁸ affecting around 20% of the population and with 40% the risk of developing lethal CVD.⁹ For instance, aortic aneurysms affect 4.8% of the overall population and have a 70% mortality rate after rupture.¹⁰ Recent evidence suggests that aneurysms rupture is most likely to occur in stiffer arteries.¹¹ Therefore, evaluating the progression of AWS in aortic aneurysms would allow to develop early indicators of disease severity evolution before the onset of irreversible pathological events.

Unfortunately, the hemodynamic quantities assessable by means of clinical measurement (such as medical imaging) are limited and the techniques not always non-invasive. However, mathematical approaches for formulating and solving inverse problems in hemodynamics have the capacity to extract and estimate unobserved quantities of interest from (non-invasive) hemodynamic measurements and can, for instance, reconstruct incomplete datasets and enrich or denoise measurements. This work aims at providing a wide overview of the recent advances of mathematical formulations and computational methods concerned with inverse problems in hemodynamics. The formulation and solution of these problems will depend heavily on the quantity of interest and the type of data available.

To the best of our knowledge, the first review on image-based blood flow modeling was published in 2010 by Taylor and Steinman.¹² This review, however, did not consider model personalization in terms of solutions of inverse problems but rather of direct imposition of measurements (e.g., geometry, flow rate and pressure forms) in the model. The reason is that, at that time, inverse hemodynamics was just starting as a research field. The first work reviewing and summarizing results concerning variational data assimilation (for both boundary conditions in stationary flows and stiffness in fluid–solid interaction was reported in 2012 by D'Elia et al.¹³ which was later extended by Veneziani and Vergara.¹⁴ A year later, Marsden¹⁵ reviewed optimization problems and approaches in cardiovascular flows, with a part devoted to the estimation of properties of numerical flow models from data. The most recent effort in reviewing inverse hemodynamics methods and applications was carried out in 2017 and 2019 by Quarteroni et al.^{16,17} Here, detailed descriptions of the methodologies for variational and sequential parameter estimation approaches are given, while the review of published research proceeds until 2014. Since then, an important number of interesting publications have appeared leading to major advances in the field in terms of methodologies and applications. Therefore, the aim of this work is to present an updated overview of the current state-of-the-art approaches from a *modeler*'s perspective. In particular, we analyze and discuss the outcomes of each of the works in detail with the aim of revealing the research gaps requiring future scientific developments.

The remainder article is organized as follows:

- Section 2 presents the modeling assumptions and equations used for blood flow modeling.
- Section 3 summarizes the measurement techniques typically used to study blood flows.
- Section 4 reviews the methods for directly computing pressure gradients from full field velocity measurements.
- Section 5 gives a background on the mathematical methods used to solve inverse problems in hemodynamics.
- Section 6 reviews the research on the estimation of distributed boundary conditions in rigid wall models.
- Section 7 reviews the research on the estimation of reduced-order model parameters serving as boundary conditions in rigid wall models.
- Section 8 reviews the research on the estimation of material properties in fluid-structure interaction models.
- Section 9 reviews the research on compensating errors in the computational geometry.
- Section 10 reviews the research on state estimation concerned with errors in the initial condition in both rigid wall and fluid-structure interaction models.
- Section 11 briefly lists works on further related topics which are not exhaustively covered in this article.

2 | BLOOD FLOW MODELS

2.1 | Full-dimensional description

Blood is a suspension of formed elements (i.e., red and white blood cells, platelets) in plasma.¹⁸ In hemodynamics—the macroscopic description of the dynamics of blood flow through the vessels—blood is considered a continuous single-phase fluid (under the continuum hypothesis¹⁹). Blood acts as a non-Newtonian fluid with viscoelastic behavior,

Under the continuum hypothesis, blood flow is assumed to satisfy linear momentum, angular momentum and mass conservation. These physical laws are mathematically represented by the incompressible Navier–Stokes equations:

$$\rho \frac{\partial \boldsymbol{u}}{\partial t} + \rho(\boldsymbol{u} \cdot \nabla) \boldsymbol{u} + \nabla p - \nabla \cdot \boldsymbol{\tau} = \boldsymbol{0}$$

$$\nabla \cdot \boldsymbol{u} = \boldsymbol{0}$$
(1)

with the velocity vector $\mathbf{u}: \Omega \times T \mapsto \mathbb{R}^3$, the pressure $p: \Omega \times T \mapsto \mathbb{R}$, in a spatial domain Ω (a subset of the blood vessel of interest) and a time interval *T*, and neglecting the gravitational force. The constant $\rho > 0$ denotes then mass density of the fluid, and τ denotes the viscous stress tensor and is determined by a constitutive equation modeling the shear behavior of blood. Classical models accurately describing the non-Newtonian rheology of blood are the Casson and the Carreau-Yasuda models.^{20,21} At high shear rates and moderate-to-high Reynolds numbers blood behaves approximately as a Newtonian fluid.^{22,23} It is often assumed that such conditions exist in the flow through large vessels.²⁴ Under the assumption of a Newtonian fluid, the viscous stress tensor becomes

$$\boldsymbol{\tau} = \boldsymbol{\mu} \Big(\nabla \boldsymbol{u} + (\nabla \boldsymbol{u})^{\mathrm{T}} \Big)$$
⁽²⁾

with the constant dynamic viscosity, μ . When this assumption is acceptable is a question of on-going debate and it has been shown to be inaccurate in some situations.²¹ This assumptions is the most used in inverse problems described later.

In order to solve Equation (1), it needs to be complemented with appropriate boundary and initial conditions \boldsymbol{u}_{0} (t_{0}) = \boldsymbol{u}_{0} . So-called Dirichlet boundary conditions impose a given velocity profile \boldsymbol{u}_{D} on the whole or a part of the boundary $\Gamma_{D} \subset \partial \Omega$, and Neumann boundary conditions specify the normal stress vector $\boldsymbol{g}_{N} = \boldsymbol{\tau}\boldsymbol{n} - p\boldsymbol{n}$ (\boldsymbol{n} is the outward normal vector) on the remainder of the boundary $\Gamma_{N} = \partial \Omega \setminus \Gamma_{D}$. In order to ensure well-posedness of the continuum model, so-called *backflow stabilizations* may be added to Neumann boundaries, at the cost of perturbing imposed value of the boundary condition.²⁵

An important aspect to mention is that in Equation (1) the pressure field is defined up to a time-dependent spatial constant since $\nabla p(\mathbf{x}, t) = \nabla (p(\mathbf{x}, t) + \delta(t))$. For pure Dirichlet problems ($\Gamma_D = \partial \Omega$), this constant remains undefined and should be fixed numerically. Neumann boundary conditions on a part of the boundary uniquely determine the pressure constant, but this value can be seen as arbitrarily user-defined; indeed the velocity field depends only on the gradient of the pressure and not on its absolute value. This has the consequence that inverse problems based on the incompressible Navier–Stokes equations or some derived simplification, can only uniquely determine *a relative pressure field* with respect to some reference value, for any instant of time.

The arterial system takes an active part in continuously delivering blood at high pressure to the peripheral vasculature.^{26,27} In particular, the large arteries deform elastically under increasing blood pressure during systole and act as a reservoir ("Windkessel") storing blood which is ejected during diastole. Also the long muscular arteries and arterioles actively control the blood propagation to tissue and organs by different mechanisms.^{26,27} In a mechanistic setting, the elastic deformation, as a response to forces exerted by the blood flow on the vessel wall, can be accounted for by coupling the Navier–Stokes Equations (1) with the equations of motion of (non)linear solid mechanics:

$$\rho_s \frac{\partial^2}{\partial t^2} \boldsymbol{d}_s - \nabla \cdot \boldsymbol{\sigma}_s(\boldsymbol{d}_s, \boldsymbol{\theta}_s) = \boldsymbol{0}$$
(3)

with ρ_s the density of the solid tissue and $d_s : \Omega_s \times T \mapsto \mathbb{R}^d$ denoting the displacement vector of material points contained in the solid domain Ω_s . The Cauchy stress tensor, σ_s , the distribution of stresses inside the material, depends, in the general case, nonlinearly not only on the displacement d_s (and its derivatives), but also on parameters θ_s of a constitutive law characterizing the material.²⁸ Biological tissues are often modeled as hyperelastic materials with phenomenologically derived constitutive laws. These models establish a nonlinear relationship between deformation and stresses via a strain-energy function describing the material properties, which can include inhomogeneity and anisotropy, respecting,

3 of 42

WILEY

for example, the layered structure and fiber orientation of the arteries.^{29,30} The equation of motion (3) requires initial conditions for the displacement, $d_s(t_0)$, and for the velocity field, $d_s(t_0)$. Boundary conditions in arterial models impose combinations of Dirichlet ($d_s = g_{s,D}$) and Neumann boundary conditions ($\sigma_s n = g_{s,N}$) in the radial and tangential directions, or Robin-type conditions with elastic and viscoelastic coefficients α_s and β_s in order to represent the effect of the surrounding organs,³¹

$$\boldsymbol{\sigma}_{s}\boldsymbol{n}=-\alpha_{s}\boldsymbol{d}_{s}-\beta_{s}\boldsymbol{d}_{s}-p_{0}\boldsymbol{n},$$

where p_0 is the external pressure.

WILEY

4 of 42

2.2 | Geometrically reduced-order models

Full-scale hemodynamic simulations of the complete vasculature will remain unfeasible for the foreseeable future.³² Therefore, detailed 3D computations are restricted to particular regions of interest of the cardiovascular system (for instance the aortic arch, heart valves, aneurysms or the carotid artery exhibit strong 3D flow phenomena) and have to be carried out within truncated computational domains. Geometrically reduced-order models (GROM) can be derived with the aim to deliver realistic boundary conditions for full-dimensional models, accounting for the effects of the omitted parts of the vascular system, or in order to construct a surrogate model of the full-dimensional system, which can be solved more easily.

Lumped-parameter models (LPM) group the spatially distributed vascular system into the so-called compartments, over which the conservation laws are averaged to obtain ordinary differential equations (ODE) modeling the bulk hemodynamics, for example, in terms of the flow rate and the pressure. LPM have been used to simulate the vascular system via multiple coupled compartments and to model the truncated vasculature, acting as a Neumann boundary condition for the higher dimensional models (see Figure 1A).



(A) 3D+LPM multiscale model and LPM surrogate model for a patient-specific aorta, from Pant et al.³⁵ (B) 1D network of the arterial tree from Lombardi³⁶

FIGURE 1 Examples of reduced-order modeling approaches. (A) and multiscale 3D + LPM setting (left) and 0D surrogate LPM (right) for a patient-specific aorta³³ and (B) a complete 1D network of the arterial network.³⁴ Reprinted with permission

The most popular choice is the *three-element Windkessel model*, including the vessel compliance *C*, the proximal resistance R_p and the distal resistance R_d^{35} (proximal or distal refer to the interfaces of the modeled vessel situated nearer and farther from the heart). This model is given by the following ODE in terms of the distal pressure $p_d(t)$:

$$C\frac{\mathrm{d}p_d}{\mathrm{d}t} + \frac{p_d}{R_d} = Q,$$

$$R_p Q + p_d = P,$$
(4)

WILEY 5 of 42

where P and Q denote the pressure and flow rate at the proximal interface.

Modeling (parts of) the vascular system as a network of 1D compliant pipes, simplified equations of mass and momentum conservation can be derived, for example, in terms of the cross-sectional area A(x, t) and the flow rate Q(x, t),

$$\frac{\partial A}{\partial t} + \frac{\partial Q}{\partial x} = 0,$$

$$\frac{\partial Q}{\partial t} + \frac{\partial}{\partial x} \left(\alpha \frac{Q^2}{A} \right) + \frac{A}{\rho} \frac{\partial p}{\partial x} + k_r \frac{Q}{A} = 0.$$
(5)

The system is closed with a suitable constitutive law relating the internal pressure p(x, t) to the vessel wall mechanics,³² $p = \Phi(A; \dot{A}, \ddot{A}, ...)$. The constant k_r denotes the viscous resistance per tube unit length. Coupling conditions, imposing continuity of the fluxes and the total pressure need to be applied at bifurcations and discontinuities in the material properties. 1D vascular networks are often used as "standalone" surrogate models, as illustrated in Figure 1B, but sometimes coupled to full-dimensional models.

The coupling of models of different geometric dimension is referred to as geometric multiscale modeling.^{36,37} Typically, the coupling is done via

$$\boldsymbol{g}_N = -P\boldsymbol{n}$$
 on Γ , $Q = \int_{\Gamma} \boldsymbol{u} \cdot \boldsymbol{n} \mathrm{d}\Gamma$

with *n* being the normal surface vector and $\Gamma \subset \partial \Omega$ the part of the boundary where the coupling is enforced.

3 | MEASUREMENTS OF BLOOD VESSELS AND FLOW

Acquisition of hemodynamic data is required in the current clinical practice to estimate quantities of diagnostic interest, and in addition serves as input for inverse problems. This section summarizes the different types of hemodynamic data and the corresponding measurement techniques.

3.1 | Anatomy

The anatomy of the vessels is routinely used in the clinical assessment of vascular pathologies. Moreover, anatomical images provide—after segmentation and mesh generation using specialized software—the computational domain Ω used for the blood flow simulations and the inverse problems. Moreover, by segmenting, for example, the arterial wall over time one can measure a surrogate of the wall displacement d_s , which can be used in the context of inverse problems.

3.1.1 | Computerized tomography angiography

Computerized tomography (CT) measures X-ray attenuations induced by different tissues inside the body. Multiple Xray measurements taken from different angles are reconstructed into a single tomographic image. In CT angiography (CTA), a contrast media (typically iodine-based) is injected into the subject's bloodstream in order to distinguish the

6 of 42 LWILEY.

blood vessels from their surroundings.³⁸ The main advantage of CTA is its high spatial resolution, which may be submillimetric depending on the size of the region if interest. CTA is usually obtained at only one cardiac phase, since for prolonged scan times, the radiation dose can exceed the annual recommended maximum level.³⁹

However, in research settings cardiac-gated CTA images are obtained allowing to obtain cardiac and arterial wall motion. An example of a CTA image of a patient's aorta is shown in Figure 2 (left column).

3.1.2 Magnetic resonance angiography

Magnetic resonance imaging (MRI) functions by magnetically exciting hydrogen nuclei of the subject tissue and measuring the magnetization generated in response. From the characteristics of the measured signal, different types of tissues and their properties can be identified with great versatility. Manipulation of the sample by means of magnetic field gradients allows to encode, for example, spatial location, to create a 2D or 3D image, but also sensitizes the measurement to specific physical processes of interest, such as diffusion or flow.

Magnetic resonance angiography (MRA) is a variant of MRI for the context of anatomical vessel imaging, usually performed to extract the 3D blood vessel lumen, often using additional contrast media in order to alter the magnetic properties of the blood and to provide an improved contrast with respect to other tissues.⁴² An example of a MRA image of a patient's aorta is shown in Figure 2 (right column).

Due to its limited spatial resolution (around 2 mm voxel size), measuring the arterial wall dynamics is still not reliable: for instance, in large arteries deformations of the wall range between 2 mm to 4 mm, of the same order of the spatial resolution of MRA.⁴³ Additionally, noise and other imaging artifacts severely corrupt the recovery of the arterial wall motion dynamics.

3.2 **Blood flow velocity**

Currently, the clinically relevant measurement techniques to assess hemodynamic flow velocities \boldsymbol{u} are Doppler ultrasonography and Phase-Contrast MRI (PC-MRI). Both methods a non-invasive.

3.2.1 Doppler ultrasonography

Doppler ultrasonography is capable of real-time local velocity measurements along a beam or in two-dimensional (2D) planes.²⁶ It is versatile, non-invasive, free of ionizing radiation and can detect relatively small structures, such as leaflets and narrow jets.⁷ Typical values of axial resolution range from 0.3 mm to 1.5 mm depending, for example, on the ultrasound frequency. The attenuation of the ultrasound beam increases at higher frequencies, so there is a tradeoff between penetration depth, axial spatial resolution and proximity to the probe.⁴⁴ A further limitation

FIGURE 2 Anatomical images of blood vessels. Coronary arteries via CTA⁴⁰ (left) and Coarctations of the aorta via MRA after thresholding non-blood tissues⁴¹ (right). Reprinted with permission



of Doppler ultrasonography is that the maximum velocity that can be measured is limited within a certain range depending on the acquisition parameters. In that case, the velocity will be displayed opposite to the actual flow direction showing *velocity aliasing*.⁴⁴ A typical 2D Doppler ultrasonography measurement is illustrated in Figure 3 for the carotid artery.

3.2.2 | Phase-contrast magnetic resonance imaging

Information on the motion of a tissue—including blood flow—can be encoded with MRI in the phase of the signal emitted by the tissue by application of adequate magnetic gradients.^{46,47} The procedure is known as *phase contrast MRI* (PC-MRI). In time-resolved measurements of vascular flow, images have to be acquired over multiple cardiac cycles, implicitly producing phase-averaged data. Usually, 20–30 cardiac phases are considered.

2D PC-MRI is a standard and widely available technique in clinics,⁴⁸ with typical in-plane resolutions of 1.5 mm to 2 mm and through-plane resolutions of around 6 mm. The acquisition time corresponds to about one breathhold per 2D plane.

3D PC-MRI—when time-resolved, also called 4D $Flow^{49}$ —is usually acquired at a isotropic spatial resolution of 2 mm to 3.5 mm with scan durations between 10 and 20 minutes,⁵⁰ therefore preventing its widespread clinical use. However, the acceleration of 4D Flow sequences is a very active research field.^{51–53} A further limitation of 4D Flow is that the maximum resolved velocity is determined by the scanner operator, larger velocities resulting in aliasing. This upper velocity limit, the so-called *VENC*, relates to the strength and duration of the velocity encoding gradients.⁴⁹ As a consequence, a larger *VENC* also leads to higher noise intensity and iterative adaptation may be required to find an optimal value.

Figure 3 (right column) displays an example of blood flow streamline visualization of in vivo 4D Flow data.

The 4D Flow measurement protocol can also be extended to measure turbulence statistics⁵⁴⁻⁵⁷ (i.e., the Reynolds stress tensor⁵⁸).

3.3 | Pressure

As indicated in Section 2.1, pressure measurements are always relative with respect to a reference value, i.e., differences $p(\mathbf{x}_a) - p(\mathbf{x}_b)$ between two locations \mathbf{x}_a and \mathbf{x}_b .

In the clinical practice, both invasive and noninvasive approaches measure the difference between the patient's and the atmospheric pressure, making the values comparable among subjects.



FIGURE 3 Blood flow velocity measurements. Doppler measurement of the internal carotid artery⁴⁵ (left) and 4D Flow MRI of an CoA⁴¹ (right). Reprinted with permission

3.3.1 | Sphygmomanometer (non-invasive)

Monitoring the blood pressure non-invasively may be the simplest and most widely used measurement to assess cardiovascular health. That measurement device—a sphygmomanometer—is often composed of an inflatable cuff to collapse and then release the artery in a controlled fashion and a device to measure the pressure applied.⁵⁹ However, it is well known that sphygmomanometer pressure provides only an indication of the intra-lumen blood pressure, and the relation between both depends on the compliance of the whole cuff-skin-arterial-muscle system.⁶⁰

3.3.2 | Applanation tonometry (non-invasive)

Applanation tonometry flattens a small part of the eyeball under a tiny sensor to measure pulse pressure.⁶¹ It is nowadays widely used in clinical practice and research to measure arterial waveforms, in particular in the carotid and femoral arteries.⁶² Artifacts in the measurements are common, for example, due to respiration (especially in the carotid), and arterial motion requires the sensor to be stabilized by the operator.

3.3.3 | Catheterization (invasive)

As mentioned in the introduction, the functional relevance of an obstruction is usually characterized clinically by the pressure difference across it. Moreover, arterial stiffness calculations require pressure measurements—relative to the atmospheric pressure—to be comparable among patients.

The intra-arterial spatial distribution of the blood pressure can be measured by means of catheterization.⁴ This technique consists in inserting a catheter equipped with a pressure transducer into the vasculature of the patient and maneuvering it, under local anesthesia and guided by fluoroscopy, to the location of interest. Although it is the "gold standard" for pressure quantification, the invasive nature of the method is associated with a risk of complications.^{63–65}

The precision of the pressure measurements can be considered within a few mmHg. In the phantom experiments of Nolte et al.,⁴¹ the standard deviation of the subtraction of two pressure measurements was estimated to be around 5 mmHg, while in in vivo measurements it is expected to be twice as large.⁶⁶ An example of in vivo catheter measurements in an aortic coarctation (CoA) patient is shown in Figure 4B, in two locations proximal and distal to the CoA, displaying also the measurement uncertainty between cycles. In addition to the measurement imprecisions, the presence of the catheters in the blood could also alter the flow and it is has been shown in in vitro and in silico studies that it could lead to possible overestimations in the pressure measurements.⁶⁷

4 | DIRECT PRESSURE GRADIENT COMPUTATION FROM VELOCITIES

We start this review article with the inverse problem most widely used in cardiology practice: pressure gradient estimation from velocity measurements. To justify the use of more advanced methods—based on 4D Flow—we first start with the simplest approaches based on the Bernoulli equation. It is important to remark that the direct methods reviewed in this section do not require solving optimization problems. Instead, they are formulated such that a pressure gradient estimate can be calculated directly (i.e., with a single linear system solution) from the available velocity data. The methods require the fluid density and viscosity, for which literature values are assumed.

4.1 | Bernoulli-based methods

Pressure gradient estimation using the Bernoulli equation is routinely used in the clinical practice since the required flow velocity measurements such as Doppler ultrasonography or 2D PC-MRI are widely clinically available.⁶⁸ Although they are mathematically trivial, we briefly describe the Bernoulli-based approaches here for their practical relevance, in descending order of their data requirements.

When it is assumed that the pressure gradient is dominated by convective and inertial terms, the pressure difference between two points x_a are x_b along the flow trajectory y(s), s = [0, 1], is given by



FIGURE 4 4D Flow-based relative pressure estimation compared to catheter data, for a patient of CoA. (A) PPE and STE results at peak systole (pressure was arbitrarily set to zero at the end of the descending aorta [DAo]), (B) top: catheterization pressure measurements (with respect to atmosphere) in the ascending aorta (AAo) and DAo; bottom: DAo–AAo pressure difference from PPE/STE compared to catheterization. The shaded area indicates ± 2 standard deviations of the catheterization pressures over cycles. ((b) adapted and reprinted with permission from Nolte et al.⁴¹)

$$p(\boldsymbol{x}_a) - p(\boldsymbol{x}_b) \approx \frac{\rho}{2} \left(\|\boldsymbol{u}(\boldsymbol{x}_b)\|^2 - \|\boldsymbol{u}(\boldsymbol{x}_a)\|^2 \right) + \rho \int_0^1 \frac{\partial \boldsymbol{u}}{\partial t} \cdot \frac{\mathrm{d}\boldsymbol{y}}{\mathrm{d}\boldsymbol{s}}(\boldsymbol{s}) \mathrm{d}\boldsymbol{s}, \tag{6}$$

with $y(0) = x_a$ and $y(1) = x_b$. This method requires measurements along the trajectory but also knowledge of the trajectory itself, although some assumptions can be made on the latter.⁶⁹ For those reasons, Equation (6) is not widely used in clinics but provides an interesting comparison to other methods when 4D Flow is available because of its reduced computational complexity.⁷⁰

In the presence of a stenosis it is assumed that the pressure gradient ∇p is only dominated by the convective term $-\rho \boldsymbol{u} \cdot \nabla \boldsymbol{u}$. In this case it can be shown starting from Equation (1) that $p(\boldsymbol{x}_a) - p(\boldsymbol{x}_b) \approx \frac{\rho}{2} (\|\boldsymbol{u}(\boldsymbol{x}_b)\|^2 - \|\boldsymbol{u}(\boldsymbol{x}_a)\|^2)$. This computation requires measuring the velocity at two carefully chosen locations, for example, via two planes in 2D PC-MRI, which may not be aligned with the flow directions. Furthermore, it is unknown which points in both planes correspond to the same streamline, hence simply the peak velocities are selected.

Due to the aforementioned drawbacks, a simpler version is used in the clinical practice: the simplified Bernoulli equation $p(\mathbf{x}_a) - p(\mathbf{x}_b) \approx \frac{\rho}{2} ||\mathbf{u}(\mathbf{x}_b)||^2$, where \mathbf{x}_b is located in the jet exiting the obstruction and it is assumed that the velocity is much larger after the obstruction. Only one measurement is required of the jet velocity.

4.2 | Navier-Stokes-based methods

The assumption that the pressure gradient is dominated by the convective term is restrictive in compliant vessels and mild obstructions. Neither complex 3D flow patterns nor dynamic effects, shape of the obstruction, or turbulence are taken into account. As a consequence, Bernoulli methods are in many cases not appropriate to estimate realistic pressure gradients.

When information about the entire flow field in the region of interest is available, for example, 3D PC-MRI data, more sophisticated methods may be used to reconstruct fields of the relative pressure in the entire domain. These methods require a 3D representation of the vessel lumen Ω , i.e., the computational domain, via segmentation of magnitude image or from an average of the velocity images.

The available methods can be grouped into two categories, computing (1) 3D maps of the relative pressure or (2) pressure differences between predefined locations. In both settings, the pressure constant needs to be fixed arbitrarily by specifying a reference pressure.

4.2.1 | 3D Spatio-temporal pressure maps

The classical approach to compute the pressure field $p(\mathbf{x}, t)$ from given velocity data \mathbf{u} is by taking the divergence of (1), which results in the so-called pressure Poisson Equation (PPE)^{71–73}: find $p(\cdot, t) : \Omega \to \mathbb{R}$, such that

$$-\Delta p = \nabla \cdot \left(\rho \frac{\partial \boldsymbol{u}}{\partial t} + \rho(\boldsymbol{u} \cdot \nabla) \boldsymbol{u} - \mu \Delta \boldsymbol{u} \right) = \nabla \cdot R(\boldsymbol{u}) \quad \text{in } \Omega.$$
⁽⁷⁾

Here, the assumption of blood to be a Newtonian fluid with dynamic viscosity μ was made and $R(\mathbf{u})$ is shorthand notation for the velocity terms of the Navier–Stokes equation. Numerical solution of (7) requires boundary conditions for p on the entire domain boundary $\partial \Omega$. Taking the dot product of (1) with the outward unit normal vector \mathbf{n} on $\partial \Omega$ yields the Neumann boundary condition

$$\boldsymbol{n} \cdot \nabla \boldsymbol{p} = -\boldsymbol{n} \cdot \boldsymbol{R}(\boldsymbol{u}) \quad \text{on } \partial \Omega.$$
(8)

The solution of the Neumann boundary value problems (7) and (8) is defined up to a spatial constant c(t). The derivation of (7) invokes higher regularity assumptions for the pressure and the velocity, i.e., higher derivatives are required to exist for both p and u than originally in (1). As a consequence, the space of admissible functions among which the pressure is sought is a subset of the natural space.

This drawback can be circumvented by an approach proposed by Cayco and Nicolaides⁷⁴ and adopted for the context of 3D PC-MRI by Švihlová et al.,⁷⁵ by introducing a divergence-free auxiliary function $w(\cdot, t) : \Omega \to \mathbb{R}^3$, such that,

$$-\Delta \boldsymbol{w} - \nabla \boldsymbol{p} = \boldsymbol{R}(\boldsymbol{u}) \text{ in } \Omega$$

$$\nabla \cdot \boldsymbol{w} = 0 \text{ in } \Omega$$

$$\boldsymbol{w} = 0 \text{ on } \partial \Omega.$$
(9)

Due to its resemblance to an inhomogeneous Stokes equation, the method is referred to as Stokes estimator (STE). Solving Equation (9), the pressure *p* is obtained together with the auxiliary function \boldsymbol{w} , whose purpose is to regularize the ill-posed pressure gradient recovery problem. In the ideal scenario with perfect velocity measurements— $R(\boldsymbol{u})$ being truly irrotational and no numerical errors—we have $\boldsymbol{w} = 0$. Note that \boldsymbol{w} could be used to explore the quality of the 4D flow data as proposed in.⁷⁶

The PPE and STE methods were compared in numerical studies.^{75,77} In Nolte et al.,⁴¹ the STE and PPE methods were evaluated using in vitro and in vivo MRI data in comparison to ground-truth catheterization pressure measurements. These studies^{41,75,77} showed that the theoretical advantages of the STE method over the PPE method have a significant effect on the results in convection dominated flow regimes (elevated Reynolds numbers), as can be expected in clinically relevant severe stenoses. In addition, the latter study⁴¹ found an improved robustness to different segmentations of the STE with respect to PPE. Figure 4A shows STE and PPE relative pressure maps computed from 4D Flow data of a CoA patient. The larger pressure gradient obtained with the STE method is in closer agreement with catheterization data, as shown in Figure 4B.

While the original PPE method neglects the diffusive term in (7), a modification was recently proposed accounting for viscous effects.²⁷ A theoretical analysis presented in Araya et al.⁷⁸ indicated an improved accuracy compared to the standard method. The authors of the present paper performed preliminary computations on the phantom data of Nolte et al.⁴¹ including the viscous effects in the PPE, and the differences with the standard PPE appeared to be negligible. Further improvements to the treatment of the viscous effects were reported by Pacheco,⁷⁹ where additionally the sensitivity of different treatments of the convective effects to noise in the data was studied.

4.2.2 | Pressure differences over predefined locations

Other approaches have been proposed for the computation of averaged pressure differences between two predefined cross-sections of a blood vessel from 3D velocity data.

The first reported approach is the so called work-energy relative pressure (WERP) method derived from the energy balance of Equation (1). Computationally, it only requires evaluating integrals of the data over the lumen of interest without solving a linear system.⁸⁰ WERP showed increased accuracy with respect to the PPE,⁸⁰ but less accurate than the STE.⁷⁷ Bertoglio et al.⁷⁷ also proved that WERP may have a considerable bias when noise in the data is present. A simplification of the WERP method was presented in Donati et al.⁸¹ (Simplified Advective WERP, or SAW) by taking only the boundary integral of the advective energy where the maximal velocity is present, therefore requiring only 2D PC-MRI data.

The energy balance in WERP is obtained by taking the scalar product of (1) with the flow velocity itself. However, other functions could be used, satisfying some properties, for instance, the velocity of a Stokes flow in the same domain. The testing Stokes flow will depend on the chosen planes. Two variants of such approach have been proposed: the integral momentum relative pressure (IMRP) estimator⁷⁷ and the virtual WERP (vWERP) method.⁸² The difference lies in the treatment of the convective terms. The vWERP method provided an increased accuracy compared to PPE and IMRP, with respect to catheter measurements in aortic flows,⁸² and obtained highly accurate estimates of the intraventricular pressure gradient using in silico data.⁷⁰ Moreover, while the WERP method requires the vessel section of interest not to include any bifurcations (such as the supra-aortic branches), this is not the case in the vWERP and IMRP methods. The vWERP was also recently applied to the estimation of pressure gradients in the brain vasculature,⁸³ where it was found that the accuracy depends on the spatial image resolution in those small vessels. Another recent study used the vWERP for the clinical assessment of hemodynamics in dilated cardiomyopathy.⁸⁴

4.2.3 | Inclusion of turbulent effects

Some of the pressure estimation methods haven been extended for turbulence by including measurements of the Reynolds' stresses (corresponding to the covariance of the turbulent velocity), basing the methods on the Reynolds-averaged Navier–Stokes equations (RANS)⁵⁸ rather than (1). RANS describe the evolution of the statistical mean flow, depending on the turbulent fluctuations in terms of the Reynolds' stresses. The methods differ from their non-turbulent versions in that they include the Reynolds' stresses as obtained with the extended 4D Flow sequence, described in the previous section. Turbulence extensions were studied for the PPE method,⁸⁵ and the WERP and vWERP methods.⁸⁶ Marlevi et al.⁸⁶ compared these methods for both in silico and in vivo data, showing the vWERP's extension (denoted vWERP-t) to be the most accurate.

So far, no studies have been conducted extending the STE method by turbulence effects. Given measurements of the Reynolds' stresses, the method can be extended in straight-forward manner, like the PPE method, by using the RANS equations instead of the Navier–Stokes equations.

4.3 | Discussion and perspectives

Accuracy comparisons between STE and vWERP—the two best performing and validated methods so far—using in vitro or in vivo data sets have not yet been performed, in particular including Reynold's stress measurements. In terms of the method of choice, approaches computing pressure differences over predefined locations are computationally cheaper compared with computing pressure maps with PPE or STE methods. However, the former need to define the position and orientation of the planes in the lumen before the pressure difference computation, while the latter allows to explore the whole pressure field first.

One particularly attractive clinical application of 4D Flow based pressure map estimators is pediatric pathologies. In many of these diseases, the flows are very slow and therefore the circulation works at low pressures. In these cases, pressures are of order of a few mmHg but catheters have measurement errors of comparable magnitude. Moreover, the location of the maximal pressure gradient is unknown and therefore probing the whole pressure field from 4D Flow data could lead to more accurate diagnosis of the pathology.

A common measure in the clinical practice are the so-called *peak-to-peak* pressure differences, which compare the difference in the pressure maxima registered at different locations during the complete cardiac cycle, thus taking into account time shifts due to the vessel elasticity. Usually, clinical reports of a patient's intra-vessel pressure will be based on this quantity, and not the instantaneous pressure difference as computed by the approaches discussed in this section. An interesting and challenging research line, and of great clinical relevance, is the correct transfer from sphygmomanometer pressure measurements to blood lumen arterial pressures. This would allow fully non-invasive peak-to-peak pressure measurements. In 4D flow-based methods, the spatio-temporal distributed data also allows to distinguish among the different contributions to the pressure gradient (inertial, advective and viscous) which could be explored clinically as markers of stenosis severity.

5 | DATA ASSIMILATION METHODS

In this section, we will review modeling and numerical approaches that aim to estimate the following information of blood flow models by measurements of u, p, d_s :

- Boundary conditions for the flow dynamics: (a) in-/outflow velocity profiles u_D, (b) the normal stress g_N, and/or (c) parameters of reduced-order models R_p, R_d and/or C.
- Elastic properties of the arterial walls θ_s and effects of surrounding organs α_s .
- Compensating for errors in the initial condition $\mathbf{u}(t_0), \mathbf{d}_s(t_0), \mathbf{d}_s(t_0)$ and the computational geometry Ω .

This section will review the mathematical approaches which have been applied to perform these estimations. In the remainder of the article, the literature employing these methods in problems of blood flows will be reviewed.

Importantly, the optimized models lead to complete hemodynamic characterizations from partial (e.g., 2D) measurements. This is an important advantage over direct pressure estimation methods reviewed in Section 4.

5.1 | General mathematical formulation

The model is a system of PDEs that describes the blood flow dynamics as discussed in Section 2. Let us introduce the following short-hand notation for the space semi-discrete numerical model, representing a differential algebraic equation (DAE),

$$\dot{X} = \mathcal{A}(X,\theta),\tag{10}$$

where $\mathcal{A} : \mathbb{R}^n \times \mathbb{R}^p \mapsto \mathbb{R}^n$ is the model operator and $X(t) \in \mathbb{R}^n$, $t_0 \le t \le T$, denotes the model state with *n* degrees of freedom and an initial condition $X(t_0)$. Depending on the effects represented in the model, it may include the fluid velocity, the solid displacements and velocities and/or the state variables of the reduced-order models. The uncertain physical model parameters, such as constants pertaining to the boundary conditions or constitutive relations, are summarized in the parameter vector $\theta \in \mathbb{R}^p$. Parameters considered certain are implicitly included in the operator \mathcal{A} .

A time discretization of the DAE (10) reads for the *n*th time step t_n , n = 1, ..., N, $t_N = T$, with the initial condition $X_0 = X(t_0)$,

$$X_n = \mathcal{A}_n(X_{n-1}, \theta). \tag{11}$$

We additionally assume that measurements, $Z_n \in \mathbb{R}^m$, are related to the state via the observation operator (possibly time-dependent), $\mathscr{H}_n : \mathbb{R}^n \mapsto \mathbb{R}^m$, in a quasi-static fashion such that

$$Z_n = \mathscr{H}_n(X_n) + \zeta_n,\tag{12}$$

where $\zeta_n \in \mathbb{R}^m$ represents measurement errors, such as noise. This relationship allows *partial* measurements, or measurements of derived quantities of the state, to be used to estimate the state and/or model parameters.

WILEY 13 of 42

We assume now that the measurement noise ζ_n follows a normal distribution with zero mean and a covariance matrix W_n . It is further assumed a priori (i.e., without any prior knowledge on the measurements), that the knowledge on the initial condition X_0 and the parameters θ is given by normal random variables with mean values \hat{X}_0 , θ_0 and covariances C_0 and P_0 , respectively. Let us summarize the unknowns X_0 and θ in the control vector, $\phi := (X_0, \theta)$, and the corresponding initial guess and its covariance as $\phi_0 := (\hat{X}_0, \theta_0)$ and Q_0 : = diag(C_0, P_0). The adoption of a *Bayesian estimation approach*—i.e., maximizing the probability that the model observes the data given prior information—defines the following cost function weighting the discrepancy between the model state and the observations and the prior knowledge:

$$J(\phi) = \sum_{n=0}^{N} \|Z_n - \mathscr{H}_n(X_n)\|_{W_n^{-1}}^2 + \|\phi - \phi_0\|_{Q_0^{-1}}^2.$$
(13)

Evaluation of *J* in a particular value of ϕ implies solving (11) to obtain X_n for the given ϕ . Data assimilation seeks the optimal trajectory of the state X_n by solving the minimization problem

$$\operatorname*{arg\,min}_{\phi} J(\phi). \tag{14}$$

The solution process of Problem (14) is facilitated by the second term in the cost function, usually referred to as *Tikhonov regularization*. Imposing additional desired properties on the solution—for instance, low energy, smoothness or, as in (13), prior knowledge—regularization is crucial when problems are ill-posed (e.g., noisy data and/or nonlinear models).

There are two main families of approaches to solve Problem (14): variational or sequential data assimilation. Both approaches have been applied to inverse problems in hemodynamics and shall be outlined in the following sections.

5.2 | Variational data assimilation

Efficient solution of the minimization problem (14) with, for example, quasi-Newton optimization algorithms, requires the gradient of the cost function *J* with respect to the control variables ϕ . Explicit approximation of the gradient with finite differences is excessively expensive in practice, since it scales linearly with the dimension of the control variables. In contrast, the *adjoint method* provides an elegant and efficient method to compute the gradient of *J*,^{87,88} at a cost basically independent of the dimension of the control space.⁸⁹

For the purpose of illustrating the adjoint method, we rewrite the PDE-constraint (10) in an abstract form.⁹⁰

$$F(X,\phi) = 0, \tag{15}$$

where ϕ represents all controls and the state *X* can be seen as the block-structured vector containing the unknowns at all times (i.e., $X = [X_1, X_2, ij, X_N]$). The boundary conditions are assumed to be included in *F*. We seek the gradient of *J* with respect to ϕ , in order to minimize the cost function *J* with constraint (15). The gradient, applying the chain rule, reads

$$\frac{\mathrm{d}J}{\mathrm{d}\phi} = \frac{\partial J}{\partial X} \frac{\mathrm{d}X}{\mathrm{d}\phi} + \frac{\partial J}{\partial\phi},\tag{16}$$

where partial derivatives $\frac{\partial I}{\partial X}$ and $\frac{\partial I}{\partial \phi}$ are easily obtained analytically. In contrast, computation of the so-called sensitivities $\frac{dX}{d\phi}$, for example, by finite difference approximations, is extremely costly and generally impractical. Differentiating the model (15), an expression can be found for the sensitivity (assuming $\frac{\partial F}{\partial X}$ is invertible),

$$\frac{\partial F}{\partial X}\frac{\mathrm{d}X}{\mathrm{d}\phi} + \frac{\partial F}{\partial\phi} = 0 \quad \Leftrightarrow \quad \frac{\mathrm{d}X}{\mathrm{d}\phi} = -\left(\frac{\partial F}{\partial X}\right)^{-1}\frac{\partial F}{\partial\phi},\tag{17}$$

^{14 of 42} WILEY

NOLTE AND BERTOGLIO

where, again, the partial derivatives of F can usually be determined analytically in straight forward manner, and where $\frac{\partial F}{\partial X}$ represents the Jacobian of the forward model. Substituting (17) in (16) yields

$$\frac{\mathrm{d}J}{\mathrm{d}\phi} = -\frac{\partial J}{\partial X} \left(\frac{\partial F}{\partial X}\right)^{-1} \frac{\partial F}{\partial \phi} + \frac{\partial J}{\partial \phi}$$

or, equivalently,

$$\frac{\mathrm{d}J}{\mathrm{d}\phi} = \lambda^{\mathrm{T}} \frac{\partial F}{\partial \phi} + \frac{\partial J}{\partial \phi},\tag{18}$$

introducing the adjoint state, λ , satisfying

$$\left(\frac{\partial F}{\partial X}\right)^{\mathrm{T}} \lambda = -\left(\frac{\partial J}{\partial X}\right)^{\mathrm{T}}.$$
(19)

Equation (19) is the adjoint equation of the PDE-constrained optimization problem. Accordingly, the gradient $\frac{d}{d\phi}$ is determined by solving the adjoint Equation (19) which depends on the forward solution X, and easy to compute partial derivatives of the model and cost function. A typical adjoint-based optimization procedure is given in Algorithm 1.

The structure of the adjoint equation corresponds to a linearization of the forward problem. Its linearity renders the solution relatively simple, with a cost comparable to or less than that of the forward problem. For time-dependent problems, however, the adjoint Equation (19) runs backward in time, starting with a terminal condition at t = T, to t_0 . At every time step, the corresponding forward solution is required, hence its entire trajectory needs to be stored. This fact leads to substantial memory requirements which are often prohibitive in practice. A reduction in memory requirements at the cost of increased computation time is achieved by splitting the time interval into subintervals and storing forward solution checkpoints only at the start of every interval. The adjoint computation proceeds subinterval by subinterval, recomputing the forward trajectory from the checkpoints when needed, and discarding old checkpoints when these become unnecessary. Checkpointing schedules⁹¹ determine the optimal number of checkpoints in order to balance the tradeoff between storage needs and computation time according to the available resources and requirements.

Practical implementations have to choose if the adjoint equation is derived analytically, prior to discretization, or if a discrete adjoint is constructed from the discretized forward problem. The former approach is referred to as optimizethen-discretize (OD), the latter as discretize-then-optimize (DO). Both approaches have advantages and are employed successfully in practice (see, for example, the discussion in Gunzburger⁹²). For instance, continuous adjoint equations (and cost function gradient) allow to use different and specific computational facilitators (e.g., stabilization schemes) for the forward and adjoint problems. However, gradients obtained with the OD approach are inconsistent with the discretized cost function and forward problem, which can cause optimization algorithms to fail. The DO approach avoids this issue and furthermore allows the adjoint to be derived by means of automatic/algorithmic differentiation tools (with their own benefits and pitfalls),^{90,93} and is often preferred in practice. In general, adjoint-based variational approaches are

ALGORITHM 1 General adjoint-based optimization algorithm

Given an initial guess ϕ^0 , repeat for k = 0, ij, N_{max} , until convergence:

- 1. Compute forward solution X^k for controls ϕ^k [Equation (10)]
- 2. Compute adjoint solution λ^k [Equation (19)]
- 3. Compute gradient $\frac{dJ}{d\phi} 0.0001 p t_{\phi^k}$ from λ^k [Equation (16)] 4. Decrease the value of J using direction $\frac{dJ}{d\phi} 0.0001 p t_{\phi^k}$ (e.g., using some quasi-Newton method) to obtain new controls $\hat{\phi}_{1}$

5. Set
$$\phi^{k+1} \coloneqq \widehat{\phi}^k$$

popular when the dimension of the control space is large, as is the case when estimating distributed boundary conditions and/or the initial conditions.

5.3 | Sequential data assimilation

The adjoint-based variational data assimilation approach fits the entire trajectory of X_n to the ensemble of observations. Sequential DA instead uses a recursive procedure in which all observations are assimilated during one forward time integration of the model, once they are "encountered." Specifically, at each time t_k sequential methods find the estimate based on the cost function (13) with the upper limit of the sum set to the current iteration, N = k. In this way, at each assimilation instant, the new information is added to the previous 'knowledge' to improve the model predictions. In contrast to adjoint variational DA, the sequential approach considers at a time t_k all previously gathered observations, but not 'future' observations.

5.3.1 | Kalman filtering

Kalman filtering is the classic approach of sequential data assimilation⁹⁴ and consists in an a priori prediction step (or forecast in DA jargon) by the numerical model and an a posteriori correction step (or analysis), incorporating the observations into the state estimation. For its introduction, let us consider a linear model for the discrete problem (11) with operator A_n at time t_n for an unknown true state X_n^t to be estimated:

$$X_n^t = A_n X_{n-1}^t + B_n. (20)$$

 B_n is the vector containing boundary condition and source terms. For simplicity, we assume that unknown parameters θ are appended to the state vector X_n , and that A_n is extended accordingly by an identity matrix block, such that a pseudo-propagation law $\theta_n = \theta_{n-1}$ is obtained. We introduce an a priori prediction X_n of the unknown true state X_n^t and an a posteriori correction \hat{X}_n , the estimation errors $\epsilon_n = X_n - X_n^t$ and $\hat{\epsilon}_n = \hat{X}_n - X_n^t$, and the corresponding error covariance matrices,

$$P_n = \mathbb{E}[\epsilon_n \epsilon_n^{\mathrm{T}}], \quad \hat{P}_n = \mathbb{E}[\hat{\epsilon}_n \hat{\epsilon}_n^{\mathrm{T}}]$$

Given a correction \hat{X}_{n-1} at time t_{n-1} , the a priori prediction at the next time step t_n is computed by propagating the correction with the forward model,

$$X_n = A_n \hat{X}_{n-1} + B_n.$$
 (21)

The Kalman filter finds the optimal a posteriori correction by balancing the model prediction with the observation data,

$$\hat{X}_n = X_n + K_n (Z_n - H_n X_n), \tag{22}$$

where H_n is the observation operator, also assumed linear, Z_n denotes the measurements. The weighting between model predictions and observations, K_n , referred to as the Kalman gain, is defined such that the a posteriori error estimate variances $(tr\hat{P}_n)$ are minimized,

$$K_{n} = P_{n}H_{n}^{\mathrm{T}}(W_{n} + H_{n}P_{n}H_{n}^{\mathrm{T}})^{-1}, \qquad (23)$$

where W_n denotes the covariance matrix of the measurement error. For the covariance matrices, the prediction step is given by the recursion formula

$$P_n = A_n \hat{P}_{n-1} A_n^{\mathrm{T}},\tag{24}$$

with \hat{P}_0 a given initial condition. The correction error covariance is computed with

^{16 of 42} WILEY-

$$\hat{P}_n = (I - K_n H_n) P_n. \tag{25}$$

Summarizing, a time step of the Kalman filter computes in a prediction step priors of the state estimate and the covariance matrix (21), (24) followed by a correction step consisting in the computation of the Kalman gain (23) and updates of the state estimate and covariance (22), (25). For a complete and instructive derivation, the interested reader is referred to Asch et al.⁹⁵

In the linear case, the Kalman gain contains the inverse of the Hessian of the cost function, and therefore provides an exact approach to solve the minimization problem but in a recursive fashion over time. In the nonlinear case, different approaches for computing K_n give rise to variants of the Kalman filter. The Extended Kalman filter is the direct adaptation of the linear version to nonlinear problems by linearization. It is impractical for most problems, since it requires a tangent linear model, operates on the full, dense covariance matrix and requires the nonlinearities to be weak. The Ensemble Kalman Filter⁹⁶ (EnKF) and its variations⁹⁷ and the Unscented Kalman Filter⁹⁸ (UKF) use instead ensembles of state perturbations (referred to as "particles") to approximate the error statistics (state estimate mean and covariance). They differ mainly by using deterministic or stochastic particles, respectively, and the latter employing a low-rank approximation of the covariance matrix, thus reducing the number of required particles.

Practical advantages with respect to variational DA are that storage of the state is not required for all times and that gradients of the cost function *J* are approximated using "derivative-free" approaches. The price to pay is that the Kalman gain matrix K_n —of the size of the dimension of the uncertain parameters and/or initial condition—is not sparse. Therefore, using Kalman filter methods for estimating $X(t_0)$ in realistic hemodynamic problems is computationally expensive if no assumptions are made to severely reduce the problem size. Also, the large number of particles required (for instance, 50–100 for the EnKF)⁹⁹ results in a high demand in CPU time, since for each particle one independent forward problem has to be solved. These particle forward problems can be solved simultaneously on a parallel computer if enough computational resources are available.

5.3.2 | Reduced-order Kalman filtering (perfect knowledge on $X(t_0)$)

Data assimilation problems can be greatly simplified by neglecting the uncertainty in the initial condition of the state (number of unknowns of the order 10^5 to 10^7) and only considering uncertainties in the parameters (typically dozens or less), describing, i.e., boundary conditions and material properties. Both variational and sequential DA methods are applicable to the resulting parameter estimation problem. For small numbers of parameters, the sequential approach offers the advantages of computational efficiency due to the recursivity avoiding the storage of the state, but also due to its implementational simplicity without the need of deriving or implementing adjoints.

A reduced-order version of the UKF for parameter estimation was presented in Moireau and Chapelle,¹⁰⁰ referred to as the reduced-order unscented Kalman filter (ROUKF). It assumes that the uncertainties at the initial time are of low rank, for instance concentrated on the physical parameter θ to be estimated. Hence, the number of particles it employs is the number of parameters to be estimated plus one, which renders the problem very tractable since the solutions of the particles can be fully parallelized and the matrices involved in the Kalman filtering approach remain low rank. This approach has been extensively applied in blood flows as will be discussed in the following sections. The ROUKF method is presented in detail in Algorithm 2.

State observers (perfect knowledge on θ)

When dealing with uncertainties in the initial condition only, alternative tractable filtering approaches can be formulated by modifying the system dynamics as:¹⁰³

ALGORITHM 2 Factorized Reduced-order Unscented Kalman Filter (ROUKF) after Moireau and Chapelle.¹⁰⁰

Problem: Consider a parameter estimation problem (11)–(14) with nonlinear model and observation operators. The initial condition $\hat{X}_0 \in \mathbb{R}^r$ is known exactly and p unknown model parameters $\theta \in \mathbb{R}^p$ are sought for which an initial guess $\hat{\theta}_0$ and, as a measure of its uncertainty, the covariance $P_0 \in \mathbb{R}^{p \times p}$ are given.

Definitions: X_n denotes a prior prediction of the state and \hat{X}_n a posterior correction. Let $[Y^{(*)}]$ denote the matrix with the column-wise collection of vectors $Y^{(1)}, Y^{(2)}, \dots$ Define the *simplex sigma-points* $I^{(i)}, \dots, I^{(p+1)} \in \mathbb{R}^p$ given such that $[I^{(*)}] \equiv [I_p^{(*)}] \in \mathbb{R}^{p \times (p+1)}$ is computed recursively as^{101,102}

$$\left[I_1^{(*)}\right] = \left[-\frac{1}{\sqrt{2\alpha}} \quad \frac{1}{\sqrt{2\alpha}}\right], \ \alpha = \frac{1}{p+1}$$

and

$$\begin{bmatrix} I_d^{(*)} \end{bmatrix} = \begin{bmatrix} & & & 0 \\ & \begin{bmatrix} I_{d-1}^* \end{bmatrix} & \vdots \\ & & & 0 \\ \frac{1}{\sqrt{\alpha d(d+1)}} & \cdots & \frac{1}{\sqrt{\alpha d(d+1)}} & \frac{-d}{\sqrt{\alpha d(d+1)}} \end{bmatrix}, \ d = 2, ..., p \ .$$

Initialization: initialize the sensitivities as

$$L_0^{\theta} = \sqrt{P_0^{-1}} \quad \text{(Cholesky factor)}, \quad L_0^X = 0 \in \mathbb{R}^{r \times p}, \quad U_0 = P_{\alpha} \equiv \alpha \Big[I^{(*)} \Big] \Big[I^{(*)} \Big]^{\mathrm{T}} \tag{26}$$

Then compute for n = 1, ..., N:

1. Sampling: generate p + 1 particles from the current state and parameter estimates, for i = 1, ..., p + 1:

$$\begin{cases} X_{n-1}^{(i)} = \widehat{X}_{n-1} + L_{n-1}^{X} C_{n-1}^{\mathsf{T}} I^{(i)}, \\ \theta_{n-1}^{(i)} = \widehat{\theta}_{n-1} + L_{n-1}^{\theta} C_{n-1}^{\mathsf{T}} I^{(i)} \end{cases}$$
(27)

with C_{n-1} the Cholesky factor of U_{n-1}^{-1} .

2. Prediction: propagate each particle with the forward model and compute an a priori state prediction:

$$\begin{cases} X_{n}^{(i)} = \mathcal{A}_{n} \left(X_{n-1}^{(i)}, \widehat{\theta}_{n-1}^{(i)} \right), & \theta_{n}^{(i)} = \theta_{n-1}^{(i)}, & i = 1, ..., p+1 \\ X_{n} = E_{\alpha} \left(\left[X_{n}^{(*)} \right] \right) \equiv \alpha \sum_{i=1}^{p+1} X_{n}^{(i)}, & \theta_{n} = \widehat{\theta}_{n-1} \end{cases}$$
(28)

3. Correction: compute a posteriori estimates based on measurements for state and parameters:

WILEY 17 of 42

$$\begin{cases} \Gamma_{n}^{(i)} = Z_{n} - \mathscr{H}(X_{n}^{(i)}) & i = 1, ..., p + 1 \\ L_{n}^{X} = \alpha [X_{n}^{(*)}] [I^{(*)}]^{\mathrm{T}} \\ L_{n}^{\theta} = \alpha [\theta_{n}^{(*)}] [I^{(*)}]^{\mathrm{T}} \\ L_{n}^{\Gamma} = \alpha [\Gamma_{n}^{(*)}] [I^{(*)}]^{\mathrm{T}} \\ U_{n} = P_{\alpha} + (L_{n}^{\Gamma})^{\mathrm{T}} W_{n}^{-1} L_{n}^{\Gamma} \\ \widehat{X}_{n} = X_{n} - L_{n}^{X} U_{n}^{-1} (L_{n}^{\Gamma})^{\mathrm{T}} W_{n}^{-1} E_{\alpha} ([\Gamma_{n}^{(*)}]) \\ \widehat{\theta}_{n} = \theta_{n} - L_{n}^{\theta} U_{n}^{-1} (L_{n}^{\Gamma})^{\mathrm{T}} W_{n}^{-1} E_{\alpha} ([\Gamma_{n}^{(*)}]) \end{cases}$$
(29)

Note: $P_n = L_n^{\theta} U_n^{-1} (L_n^{\theta})^{\mathrm{T}}$ is an estimate of the parameter error covariance. W_n is the measurement error covariance matrix, i.e., for Gaussian white noise with fixed variance, $W_n = W = \sigma^2$.

$$\dot{\hat{X}} = \mathcal{A}(\hat{X}, \theta) + \mathcal{L}(Z - \mathcal{H}(\hat{X})), \quad \hat{X}(t_0) \neq X(t_0),$$
(30)

with \mathscr{L} a sparse operator. In practice, \mathscr{L} has to be designed for each type of measurements (e.g., displacements, velocities and/or pressures) and physics (e.g., fluid, solid or fluid–solid coupling) such that $\hat{X}(t) \to X(t)$ when $t \to \infty$, where X(t) denotes the true state.

This methodology is effective for estimating the state in presence of uncertainties in the initial guess but estimating model parameters is not possible. However, sequential data assimilation methods for parameter estimation can be combined with state observers in order to enable computationally inexpensive joint state/parameter estimation.¹⁰⁴

5.4 | Root-finding formulation

Root-finding is an approach for parameter estimation problems with measurements that are not distributed over the time interval, but either time-averages of hemodynamic quantities or, for example, terminal target values. Instead of least-squares minimization of the model-observation discrepancy using the quadratic cost-function (13), the root-finding problem reads: *find* θ *such that*

$$Z - \mathcal{H}(X(\theta)) = 0 \quad \text{with } X, \ \theta \text{ satisfying } (10), \tag{31}$$

assuming that the measurements Z and $\mathcal{H}(X)$ correspond to a specified instant of time or cycle averages (and not temporally resolved data).

Such problems are usually solved by means of (quasi-)Newton methods, requiring the Jacobian (or an approximation) with respect to θ of the residual $Z - \mathcal{H}(X)$ at every iteration of the algorithm. Due to the nature of the root-finding problem, the number of measurements has to match the number of parameters for the roots of (31) to be determined, in contrast to least-squares optimization approaches. Root-finding lacks the optimality properties, the theory and generality of optimization, but can be very cheap in the specific cases where the method is applicable and converges. Details of the application of these methods to blood flows are given in Section 7.5.

6 | ESTIMATION OF BOUNDARY CONDITION DATA

6.1 | Preliminaries

Full-dimensional hemodynamic simulations require inflow and outflow boundary conditions at the interfaces of the computational domain, accounting for the truncated vasculature. One of the most explored research topics in the field has been the estimation of distributed boundary conditions—i.e., velocity u_D or normal stress boundary data g_N —for

Problem (1) from distributed interior velocity measurements (e.g., from 2D or 3D PC-MRI) using (mainly variational) data assimilation approaches. For the sake of readability, we structure this section by classifying the literature by stationary and time-dependent settings.

6.2 | Averaged boundary data

Some of the first inverse problems in blood flows were motivated by the limited clinical availability of measurements, which consisted mainly in flow rates and mean pressures. Imposing these types of data as boundary conditions was called *defective boundary conditions* and received significant attention in the first decade of this century. This "data imposition" involves solving an inverse problem.

Formaggia et al.¹⁰⁵ treated the problem of imposing average pressure and flow rate boundary data in rigid-wall flow problems. An inverse problem was formulated with the desired boundary data, i.e., either pressure or flow rate, as "measurement" while the other quantity was used as control variable. It was proposed to solve this problem using iterative minimization algorithms with the gradient of the cost function based on adjoint equations. Additional constraints on the boundary velocity were imposed. Existence of solutions is proved for the stationary linearized case, as well as for the Navier–Stokes problem with large viscosity. The problem for the transient case was formulated in a quasi-stationary fashion for each time step. Numerical examples in transient 2D flows with one and two control boundaries exemplified the methodology, and the convergence of the numerical algorithm was proved. For the examples shown only a couple of iterations of the minimization procedure were needed. However, the convergence behavior in the presence of a larger number of control boundaries was not investigated.

Formaggia et al.¹⁰⁶ developed the extension to fluid–solid interaction for the flow rate data case, where various numerical algorithms were proposed and investigated. However, for a similar 2D test case with two boundaries, the number of iterations increased by a factor of 10 with respect to the rigid walls example previously reported.¹⁰⁵

6.3 | Distributed boundary data in stationary flows

6.3.1 | Estimation of boundary normal stress

Distributed boundary normal stress g_N were first estimated in D'Elia and Veneziani¹⁰⁷ from synthetic velocity measurements using variational data assimilation on a Stokes model, which was later extended to the Oseen and Navier–Stokes equations in D'Elia et al.¹⁰⁸ In both works, the optimal control problem was solved via the discretize-then-optimize approach and tested in 2D synthetic examples, with velocity measurements on interior slices. Conditions for the existence of a unique minimizer of the optimization problem were investigated, revealing that a Tikhonov regularization for the flow velocity allows to avoid constraints on the position of the measured data. Furthermore, it was shown that interpolating the data to different locations also had a regularizing effect, such that the Tikhonov regularization could be omitted.

Adib et al.¹⁰⁹ addressed the problem of estimating boundary normal stress—assumed constant over each outlet boundary—by means of bruteforce minimization. Results computed for real 4D Flow data from patients with intracranial aneurysm were shown, where agreements between the optimized model and data were satisfactory even if data coming from a time-dependent regime. However, the bruteforce optimization approach used scaled exponentially with the number of boundaries, so the authors recognized that this could be improved, for example, by introducing local minimization strategy.

A monolithic approach, considering the coupled state-adjoint optimality system, was formulated by Zainib et al.¹¹⁰ to estimate distributed normal stress boundary conditions from 4D Flow-type measurements. The main contribution of the work was to show that a reduction of the state-adjoint optimality system dimension of more than three orders of magnitude is possible using proper orthogonal decomposition (POD).

6.3.2 | Estimation of boundary velocity profiles

First in Tiago et al.¹¹¹ and Guerra et al.,¹¹² the velocity boundary profiles u_D of 2D fluid flow problems were optimized using a discretize-then-optimize approach, including a regularizer on the gradient of the boundary velocity. Later, the

analysis was extended to 3D, allowing to recover helical and secondary flow patterns downstream of the inlet.^{113,114} Numerical results showed that the regularizer allows to estimate u_D from flow velocity measurements only in a subset of the domain. The existence of a minimizer of the optimization problem for the case that the velocity is measured everywhere in the domain was proven in a previous publication.¹¹⁵

Koltukluoğlu and Blanco¹¹⁶ first applied a similar problem formulation to experimental MRI phantom data. A optimize-then-discretize approach was adopted for the optimization problem. Due to the "full field" nature of the measurements, the optimized model's solution turned out to be a filtered, denoised version of the measurements.

6.4 | Time-dependent flow

6.4.1 | Estimation of boundary data only

In order to enable treating the 3D time-dependent problem, Koltukluoğlu¹¹⁷ proposed to reduce the complexity of the inverse problem of velocity boundary data estimation from velocity data. At the core of this work was the assumption of a time periodic flow (reasonable for physiological blood flows) which allowed to address the problem in the frequency domain by means of an harmonic balance approach. Thus, the time-dependent problem is rewritten as a sequence of coupled stationary estimation problems for each frequency. The approach avoids the estimation of the initial condition, considerably reducing the dimension of the optimization problem and making it tractable for adjoint-based variational data assimilation. A further reduction of the size of the parameter space is achieved by selecting a sufficiently small number of frequencies (much smaller than the number of time steps). The method was assessed on real phantom 4D Flow MRI data, see Figure 5, resulting in an optimized velocity field that preserves the qualitative structure of the 4D Flow data but at a higher resolution and without visible noise.

Gaidzik et al.¹¹⁸ proposed to estimate the amplitude of parabolic velocity boundary conditions in inlet and outlet boundaries for each time instant from 4D Flow data. The inverse method was a variant of the nonlinear ensemble Kalman filter, namely the local ensemble transform Kalman filter (LETKF),⁹⁷ employing a fixed number of 25 particles. Although the Kalman filter assumes that the parameters are time constants—contradicting the pulsating flow rate the results obtained using a one inlet/one outlet 4D Flow phantom showed good agreement in independent velocity measurements (particle image velocimetry). The framework was later applied to 4D Flow data of a volunteer's Circle of Willies with multiple outlets,¹¹⁹ where the accuracy cannot be assessed since no ground truth data was available.

Finally, Töger et al.¹²⁰ estimated the Dirichlet velocity data of the pseudo-compressible Navier–Stokes equations from 4D Flow phantom and patient measurements. In order to reduce the complexity, the boundary velocity profiles were parametrized using few spatial (1 to 3) and few temporal (2 to 12) constants per control boundary, which were simultaneously estimated using a variational data assimilation approach. The authors included also spatial and temporal averaging in the observation operator \mathscr{H} as it is present in real 4D Flow data. It was shown that optimizing the boundary conditions using the whole 4D Flow data set—instead of only fitting those parameters from the boundary data—led to an increased precision compared to more accurate measurement modalities like higher resolution 2D PC-MRI. Those results are shown in Figure 6.

6.4.2 | Estimation of boundary data and initial condition

A first, rigorous application of variational DA to the time-dependent, full Navier–Stokes model was presented by Funke et al.,¹²¹ adopting a discretize-then-optimize approach to simultaneously estimate time-varying velocity boundary conditions and the uncertain initial condition from velocity measurements. In a synthetic 2D aneurysm, a systematic analysis of the sensitivity of the method with respect to data sparsity, noise, regularization parameters and observation operator (instantaneous vs. time-averaged) was conducted. The results showed an excellent agreement between reference and DA solution, and proved very robust with respect to the aforementioned factors, given sufficient observations (at least 8 spanning the cardiac cycle) and sufficiently small regularization parameters. As a proof of principle, the methodology was applied to a real 3D aneurysm using 4D Flow ex vivo measurements of a dog's vessel, resulting in more than 200,000 dimensions of the parameter space. In order to make the solution of this challenging problem feasible, relatively coarse numerical resolutions were used to reduce the computational cost. In addition, the viscosity was artificially increased in order to facilitate the numerical solution with the price of introducing a modeling error with



FIGURE 5 Velocity magnitude estimation applying spectral data assimilation to 4D Flow data of an aortic phantom.¹¹⁷ Left: optimized harmonic balance model; middle: 4D Flow data; right: harmonic balance forward simulation. Permission to be requested from Springer upon acceptance



FIGURE 6 Non-calibrated flow (A,C) compared to optimized flow simulations (B,D).¹²⁰ (A): streamlines of measured 4D Flow velocities, (B): optimized velocity streamlines, (C): flow rate at cross-section M1, measured versus computed before optimization, (D): flow rate at cross-section M1, measured versus optimized solution

respect to the observations. The results agreed qualitatively with the data. This study indicates that variational DA of realistic hemodynamic problems is within reach but emphasizes the present limitations in computational power and efficiency.

2D results of Funke et al.¹²¹ are displayed in Figure 7, showing the close agreement with the reference solution and the robustness to noise of the data assimilation procedure.

6.5 | Discussion and perspectives

Approaches based on directly estimating the distributed boundary conditions such as velocity or normal stress data allow for accurate reconstructions of the measured flow dynamics. These approaches can be seen as a way to enrich the data (e.g., high-resolution reconstructions from sparse data and full-domain reconstruction when velocity measurements are available in few locations) or to filter/denoise the data using the flow physics (i.e., from noisy 4D Flow measurements).

An open problem seems to be the proof of existence (and uniqueness) of minimizers of the optimization problem in the time-dependent case. Moreover, the reviewed works have dealt only with rigid wall problems, suggesting that the extension to fluid–structure interaction problems remains an open area of research for both theoretical analysis and computational approaches.

7 | ESTIMATION OF LUMPED PARAMETER MODELS' CONSTANTS

7.1 | Preliminaries

The optimized flow models from Section 6 deliver results for a specific flow condition. In contrast, as outlined in Section 2.2, LPMs can yield boundary conditions for flow models which account for the *properties* of the truncated vasculature, represented by model parameters like resistances and compliances. Those parameters can be estimated from data using the methods listed in Section 5, instead of estimating full boundary velocity or normal stress profiles. This considerably reduces the dimension of the optimization problem, allowing also for a reduced amount of measurements. Moreover, this has the advantage that other hemodynamic regimes and physiological conditions can be simulated afterwards, for example, exercise states.

In this section, we structure the review by the type of the methods used to solve the estimation problem, namely variational and sequential data assimilation, and root-finding approaches.

7.2 | Variational approaches

Fevola et al.¹²² proposed to estimate a single resistant model (i.e., C = 0) on a stationary Stokes problem using variational data assimilation, by means of formulating and solving the continuous adjoint equations. While the inflow profile was directly extracted from the data, measurements of flow rate and average pressure were used in the cost function. Since the resistances of all outlets were optimized simultaneously, the pressure constant of the problem was fixed by pressure discrepancy error in the cost function. It was shown that the flow distribution in the outlets of an ascending aorta obtained with a fully time-dependent Navier–Stokes model—using the resistances estimated from the stationary Stokes problem—was in close agreement with the one measured in real 4D Flow data sets.

7.3 | Sequential approaches

In Pant et al.,³³ the estimation of the LPM parameters in the boundary conditions from flow rate and pressure data was performed using a 0D surrogate models replacing the full multiscale model. The proposed method iterated between the parameter estimation in the 0D surrogate model using a unscented Kalman filter and (few) 3D model evaluations in order to ensure that the 3D model was accurately represented by the surrogate model. Sensitivity analysis of the surrogate model revealed the identifiability of (and combinations of) Windkessel parameters. Tests were shown for real data of carotid arteries and thoracic aorta, also illustrating the capability of data assimilation approaches to predict outputs in different hemodynamic conditions.

Arthurs et al.¹²³ applied the Reduced-order UKF (ROUKF) to the full time-dependent Navier–Stokes model with simplified fluid–structure interaction (FSI) effects on the wall to estimate the LPM parameters in the boundary conditions. Both time-varying flow rates—as extracted from 2D PC-MRI—and pressure forms obtained by applanation tonometry in the carotid were used as measurements in aortas using synthetic and subject's data. In particular, pressure measurements appeared to be crucial to obtain the correct parameters when errors in the pressure's initial condition



(A) Velocity magnitude snapshot (colors in logarithmic scale), observations (top row) and assimilated results (bottom row) for different signal-to-noise ratios (SNR).



(B) Velocity norm and wall shear stress (WSS) over time, optimzed vs. reference, for SNR matching (a).

FIGURE 7 Variational data assimilation results for time-dependent 2D synthetic data, modified from Funke et al.¹²¹

were present, as is the case in applications with real data. Results of the estimation of 27 Windkessel parameters are shown in Figure 8B, for a patient-specific aorta with synthetic data given at the observation planes shown in Figure 8A. The results illustrate the beneficial effect of pressure data on the parameter identification and on an accurate recovery

of the blood pressure. In the real data case where no ground truth parameters are available, parameters' time evolution during filtering showed a quite stable behavior. Moreover, the model with the optimized parameters matched the flow rate data not used for the estimation well. Both aspects were indicators of probable successful estimation results.

Garay et al.¹²⁴ developed a new cost function for the inverse problem that allows to estimate global fluid parameters when aliasing in the phase-contrast MRI data is present. The cost function is derived from the original magnetization signal model, hence accounting for the velocity encoding in the phase of the MRI image. Test cases consisted in estimating distal resistances in an aortic geometry from 2D image slices containing the through-plane velocity component without the need of segmenting the flow rate time curves. This allows to reduce the number of measured slices since the spatial distribution of the velocity improves the identifiably of the parameters. Results showed robust estimation of the resistances up to velocity encoding smaller than 30% of the maximal velocity in synthetic data sets. An example with real data set was shown confirming these results.

7.4 | Multi-fidelity global optimization

Perdikaris et al.¹²⁵ proposed to minimize the data discrepancy cost function (i.e. without Tikhonov regularization) to estimate parameters in hemodynamic models using multi-fidelity, derivative-free global optimization based on statistical concepts such as Gaussian processes. The application was the estimation of the total resistance (i.e., $R_d + R_p$) from the peak systolic pressure measurement in a 3D Navier–Stokes model using a 1D models as a surrogate. The examples showed that in the case with 2 outlets in the 3D geometry, and hence with 2 parameters to estimate, the parameters can be recovered with about three 3D and eighty-five 1D model computations. However, it remains unclear how the complexity—and in particular the total number of computations of the 3D model—would scale with the number of parameters to be estimated and/or the complexity of the measurements.

7.5 | Root-finding

Spilker and Taylor¹²⁶ estimated Windkessel model parameters for idealized examples and a patient-specific abdominal aorta, with varied numbers of outlets. Single time independent measurements were considered, such as minimum and maximum inlet pressures, pressure waveform shape indicators and MRI flow rates and flow rate split at bifurcations, representing the target values to be matched by the model in the root-finding procedure. Their approach, while efficiently delivering accurate results, required accurate initial guesses of the parameters for the quasi-Newton method (using a finite-difference approximation of the Jacobian) to succeed. Suitable initial approximations for the parameters and the Jacobian were obtained with the help of a 0D surrogate model in a prior step.

An alternative approach for determining the Jacobian for the Newton method was proposed by Ismail et al.,^{127,128} circumventing the requirement of accurate initial guesses: the Jacobian was obtained by means of the adjoint approach. In contrast to adjoint-based variational data assimilation, here the Jacobian only considered the LPM, and not the full-dimensional model, thus vastly reducing the otherwise immense computational cost. Their results showed good convergence, even for inaccurate initial guesses, for different patient-specific arteries.

Algorithmic improvements to Spilker and Taylor,¹²⁶ addressing the issue of the initial parameters, were introduced by Itu et al.,^{129,130} but only applied in a reduced-order 1D/0D multiscale approach.

Other iterative approaches have become popular in the literature due to their simplicity. For instance, Troianowski et al.¹³¹ proposed a simple fixed-point iteration to estimate the total resistances of 3-element Windkessel models coupled to the 3D Navier–Stokes equations using measurements of the flow rate splits and of pressure differences between the branches. From the total resistance, the remaining parameters can be estimated according to Spilker et al.¹³² The method was applied to a pulmonary artery, iterating simulations until the computed pressure differences match the measurements with the desired tolerance (errors of $\leq 1\%$ for the pressure gradient and flow were obtained after 3 to 5 iterations). The method was adopted in different studies of pulmonary hemodynamics.^{133,134}

Another popular fixed-point approach is that of Xiao et al.,¹³⁵ who used a surrogate model to iteratively determine the Windkessel parameters of a 3D model coupled with Windkessels at the outflow boundaries. The surrogate model replaced the 3D domain with a 1D model and was coupled to the same 0D Windkessel boundary conditions. In a tailored approach, the Windkessel's total resistance and capacitance were determined by iterations of the surrogate model, until measured pressure values were reached. The converged parameters were then used in the 3D model.



(A) Setup of patient-specific aorta, showing location of observations and Windkessel BC. Table: type of observations (Q: flow rate, P: pressure)



(B) Pressure profiles and estimated Windkessel parameters (solid lines) vs. ground truth (dashed lines), without (left column) and with (right column) pressure data

FIGURE 8 Parameter estimation results of Arthur et al.¹²³ using 9 three-element Windkessel boundary conditions in a patient-specific aorta from synthetic data. (A) Setting: geometry, observation planes and observed data (cf. table), (B): pressure (at outlets 1 and 9) and WK parameter estimates of cases A and B, with and without pressure observation, compared to the ground truth

7.6 | Discussion and perspectives

Root-finding approaches appear to have been more popular than least-squares based approaches for estimating LPM parameters in 3D-0D coupled problems. However, root-finding approaches have some drawbacks. Most importantly, the number of parameters must match the number of measurements, and therefore it is not obvious how to take into

account richer data sources (e.g., time series) and different data types in the root-finding formulation. In particular in pathological cases, there is no guarantee that the imposed target values will be reached by the model. This is further complicated by noise in the measurements. In addition, root-finding appears to be very sensitive to the initial guesses and may require special treatment at the startup, as proposed by, for example, Itu et al.¹²⁹

Least-squares formulations (solved either with variational or sequential approaches) offer better robustness to the aforementioned issues, in particular since they can handle overdetermined problems (i.e., more measurement data points than parameters to estimate), for instance when spatially and/or time-resolved flow rate/velocity and/or pressure data are available.

Exploiting the reduction of the adjoint to the Windkessel model only, in Ismail et al.¹²⁸ seems to give a reasonably accurate and computationally cheap approximation of the full adjoint. An interesting extension would be to investigate the performance of the reduced adjoint approach in the context of variational data assimilation.

Also, the fixed-point iterations between a 0D surrogate and the full 3D model in Pant et al.³³ seems to allow to reduce the number of iterations in the estimation process. An interesting but highly technically complex path to explore in future research to further reduce the number of iterations is by formulating such fixed-point iterations as root-finding problems and approximating the Jacobian using composition of operators, as for instance was done in fluid–structure interaction.¹³⁶

8 | PARAMETER ESTIMATION IN FLUID-STRUCTURE INTERACTION MODELS

8.1 | Preliminaries

Blood vessels are compliant. The relation between deformation and pressure forces is governed by the parameters of the constitutive model θ_s , which have mainly been studied ex vivo, see, for example, Sommer and Holzapfel.¹³⁷ However, it is accepted that the compliance of the artery is an indicator of the subject's cardiovascular health⁸ and therefore there is great interest of in vivo quantification of the constitutive properties.

Since the increase of pressure in systole deforms the vessel, inverse problems can be formulated for estimating the constitutive parameters of fluid–structure interaction (FSI) models from measurements of both the vessel deformation and forces. Due to the complexity of the FSI model, the initial condition is usually not estimated and the models are started "at rest": zero-velocity for both fluid and solid. Moreover, the initial displacement is computed by assuming that the geometry Ω is obtained from a segmented image loaded at diastolic pressure, and therefore one should recover zero stress configuration first.^{31,138–142} A simpler alternative is to correct the stresses from the fluid to the solid for all times by the one of the initial simulation time.^{43,143} Note, however, that such an approach allows to estimate only the relative distribution of *incremental* stiffness parameters rather than their absolute values as obtained in ex-vivo experiments.

Several challenges make the estimation of the arterial wall stiffness difficult. The wall deformation is also constrained by external tissues, for instance the pulmonary trunk in the ascending aorta and the spine in the descending aorta, which however can be represented by elastic Robin boundary conditions with a parameter α_s . The heart also imposes a complex motion pattern to the ascending aorta. Last but not least, the arterial wall deformations—in particular in the arch, descending and thoracic aorta—are also small and therefore clinical images like MRI do not provide quality measurements of the wall kinematics. While dynamic CT measurements can be more reliable, they involve larger doses of radiation and are therefore not routinely acquired in the clinical practice.

8.2 | Arterial wall stiffness

We first review the works presenting estimations of arterial wall stiffness, assuming the external tissue properties are known. The classification is made by the type of DA method used.

8.2.1 | Variational approaches

Perego et al.¹⁴⁴ introduced a variational approach, using a DO strategy, in order to estimate the Young's modulus of a linear elastic solid from displacement measurements at the fluid–solid interface. The problem was formulated as

stationary minimization approach at each time step, with the Young's modulus as the only control parameter. The existence of minimizers of the problem was proved. In the numerical algorithm, the estimated Young's modulus for each time step served as initial guess for the estimation in the next time step, hence avoiding the solution of challenging time-adjoint problems. The final estimate consisted in an average of the estimates over time. Numerical tests with synthetic noisy data in 2D showed good agreements of the ground truth with the estimated parameters in spite of the measurement noise. However, the authors suggested that solving for each time step independently—not considering the information of the previous solution—may suffer instabilities in the presence of highly noisy data.

Bertagna and Veneziani¹⁴⁵ further investigated how to accelerate the computations in the case that the solid is treated as a thin membrane. The approach was based by pre-computing a series of solutions for different Young's modulus and building a proper orthogonal decomposition (POD) basis, hence considerably accelerating the linear system solutions during the optimization. Synthetic 3D examples with three Young's modulus were presented in two geometries and different noise levels. For those cases, however, in turned out that the total number of linear solves was only slightly reduced compared with the case without POD. The authors discussed at the end different strategies on how to further accelerate the computations.

8.2.2 | Sequential approaches

The first results on sequential estimation of parameters in FSI problems were reported by Bertoglio et al.¹⁴³ The methodology used was a ROUKF for the estimation of the distribution of the Young's modulus from Lagrangian wall displacement measurements. The accuracy of the estimated parameters depended on the displacement-to-noise ratio. Additionally, it was shown that it is possible to identify the Windkessel resistance from the same measurements, when the wall stiffness is known. These results were extended in the case of an elastic tube in Arthurs et al.,¹²³ where the arterial wall stiffness together with Windkessel parameters were estimated from flow rate, pressure and wall displacement data.

The ROUKF approach was validated with both phantom and volunteer MRI data in Bertoglio et al.⁴³ In the experimental validation, a silicon rubber tube was considered, connected with mechanical emulator of the cardiovascular system (Figure 9A). An additional elastic patch was included in one of the segments for inducing a local stiffening. The FSI computational model of the cylinder was set up based on the data segmented at one time instant. Pressure measurements were used for the fluid boundary conditions. The Young's modulus along ten sub-segments of the tube was estimated from surface measurements from segmentations of dynamic multi-slice MR images. The estimated Young's modulus for the all segments matched the value obtained from a mechanical test, cf. Figure 9B. The location of the additional elastic patch showed an increased stiffness (*R*1, *R*2 in Figure 9B).

Also in Bertoglio et al.,⁴³ the ROUKF was then assessed for estimating the Mooney–Rivlin constitutive parameters distribution (five sub-regions) in a patient with repaired aortic coarctation. The measurements used were segmented dynamic anatomical MRI images and therefore the observation operator consisted in distance maps from the model to the segmented surfaces. While the model with estimated parameters followed closer the segmented surfaces, the stiffness was estimated larger in the regions where the aorta was in contact with surrounding vessels due to the absence of modeling of that contact.

8.2.3 | Some additional remarks

There exist other approaches that allow to obtain stiffness estimations of vessels in vivo. When invasive pressure catheterization and images of the deformation of the cross sections are available, localized stiffness measurements are possible, see, for example, Stalhand.¹⁴⁶ A more advanced approach consists in computing the vessel stiffness from deformation maps obtained from dynamic CT and from non-invasive sphygmomanometer pressure measurements.¹¹ The latter, however, assumes a linear material and neglects 3D deformation effects.

8.3 | Boundary tissue support

The inclusion of the external tissue effect on the aorta was introduced in Moireau et al.³¹ through the application of viscoelastic Robin boundary conditions in FSI models. The estimation of the elastic Robin parameter α_s from wall



FIGURE 9 FSI wall stiffness (Young's modulus) estimation setup and results from Bertoglio et al.⁴³ Reprinted with permission

displacement data—assuming the arterial wall stiffness is known—has been performed using first sequential and later variational approaches.

8.3.1 | Sequential approaches

In Moireau et al.¹⁴⁷ the parameter α_s in an FSI model of an aorta was estimated using ROUKF from segmented dynamic CT images. The external surface of the aortic model was divided mainly into ascending aorta, arch, supra-aortic branches, while the descending aorta was subdivided into spine, spine vicinity and opposite to spine. On each surface region a different Robin stiffness value to be estimated was assigned. Distance maps of model-to-image contours were considered to quantify the discrepancy between modeled and segmented surfaces. Here, the motion of the aortic root is of great importance and was also included in the model of the ascending aorta using a Lagrangian tracking approach. The results showed that the model including the optimized coefficients reproduced the measured arterial wall dynamics from the CT better than without taking into account the external tissue effect. Results obtained in a patient-specific setting with real data are shown in Figure 10, where a prior estimation with 4 tissue regions was further refined to 11 regions with distinct parameters, all showing stable convergence except the parameter representing the aortic arch.

8.3.2 | Variational approaches

Pozzi et al.¹⁴⁸ estimated the elastic tissue support parameter α_s for the carotid artery to emulate the effect of arteriosclerotic plaque. In order to characterize the plaque and other surrounding tissues, such as the nearby jugular vein, the elastic parameters of the boundary condition were spatially distributed and their values were estimated by minimizing the discrepancies between computed vessel displacements and reference values obtained from segmented surfaces from dynamic MRI data. The optimization procedure was performed using a derivate-free approach, hence no adjoints were implemented. The results showed that the surrogate model via the optimized Robin boundary conditions is closer to a full FSI model of the carotid plaque than a rigid walls simulation.

8.4 | Discussion and perspectives

One important open problem is the joint estimation of arterial wall stiffness and boundary support parameters. It is accepted that the bottleneck comes from the fact that standard measurements (like pure surface distances) are not enough to distinguish the effects of both parameters.



FIGURE 10 External tissue parameter identification for a patient-specific aorta with real data. Initial guesses from prior estimation with four tissue regions. Reprinted with permission from Moireau et al.¹⁴⁷

Another bottleneck is the quality of data. While MRI measurements of the arterial wall motion appeared to be poor due to imaging and flow artifacts and sometimes subvoxel motion of the wall, dynamic CT images are not used in clinics since they require about the annual recommended total radiation dose.³⁹ Including additional measurements in the stiffness estimation (e.g., flow velocities in MRI) could compensate for the poor data quality of the arterial wall displacements.

It is also important to mention that usually only one of the constitutive parameters is estimated in vivo, in contrast to the ex vivo setup. Estimating all arterial wall constitutive parameters requires measuring forces and deformations in all directions. However, in vivo only the surfaces and not the full arterial wall displacement field d_s can be tracked, and loading is dominated by pressure (normal) forces.

9 | COMPENSATING ERRORS IN THE COMPUTATIONAL DOMAIN

9.1 | Preliminaries

Most, if not all, personalized models assume that the computational domain Ω is not afflicted by measurement errors and uncertainty, but make the assumption, that the vessel wall is segmented at the correct position. However, the segmentation process used to determine the vessel contours is subject to errors caused by the limited image resolution, flow artifacts and partial volume effects.¹⁴⁹ A detailed analysis of the impact of geometric uncertainties in a stenosed anatomy on the pressure difference was carried out by Brüning et al.⁶⁶ and on wall shear stress by Perinajová.¹⁵⁰ Recently, the issue of geometric errors was investigated theoretically and error bounds were presented.¹⁵¹ All studies found that small changes in geometry cause important variations in the solution.

9.2 | Compensation using velocity measurements

Additional velocity measurements have been used to estimate boundary conditions on the part of the boundary which the vessel wall should be present, hence moving out from the homogeneous Dirichlet data assumption.

The data assimilation approach of Tiago et al.,¹¹¹ reviewed in Section 6.3, also addressed the case where the computational and the real domain do not match by estimating the non-zero Dirichlet boundary conditions on the boundary, thus approximating the vessel wall position. 2D tests with synthetic measurements showed good agreements between the reference (computed in the true domain) and the estimated solution—see Figure 11. This approach requires however full field velocity measurements (like 4D Flow).



Geometry error compensation by means of boundary velocity optimization.¹¹¹ Plots show velocity streamlines and velocity FIGURE 11 magnitude. The zero-velocity streamline is plotted in magenta in the bottom row plots, approximating the true domain. Reprinted with permission



FIGURE 12 Pressure differences across 3D synthetic stenoses over time, with optimized slip/transpiration ("slip") versus no-slip boundary conditions for different stenosis severities. Mean values with $\pm 2\sigma$ bands over 30 samples of partial noisy measurements, given at the inlet and in one interior plane, with data resolution/geometry error $\Delta_1 = 1$ mm and $\Delta_2 = 2$ mm. Reprinted from Nolte and Bertoglio¹⁵²

A simpler approach was introduced by Nolte and Bertoglio¹⁵² where it was proposed to augment the flow model calibration by slip/transpiration boundary conditions, whose parameters are then estimated using velocity measurements. This approach is exact in a developed pipe flow. For a more general flow, the slip and transpiration parameters can be assumed as space-time constants, which leads to a more tractable inverse problem than estimating the spatial velocity profile as in Tiago et al.¹¹¹ The additional advantage of the reduced dimensionality is that a smaller amount of measurements, for example, 2D slices with velocity information in one direction only, suffices to estimate those parameters and hence to compensate for the geometrical errors. Numerical tests using 3D synthetic Navier-Stokes flows showed that including an additional 2D slice allows for important increases in robustness of computed quantities like pressure drops across stenotic vessels. While the solution method was the ROUKF, a variational approach could have also been used.

Figure 12 illustrates for an idealized 3D stenosis with synthetic data the accuracy of the pressure differences across the stenosis, obtained on a domain with erroneous wall locations using no-slip and optimized slip/transpiration conditions. The optimization estimated the unknown amplitude of a pulsatile inflow and two slip/transpiration parameters, constant in time and space, from synthetic 2D MRI flow data. The agreement between the ground-truth reference solution and slip//transpiration model is significantly more accurate than standard no-slip boundary conditions.

9.3 | Discussion and perspectives

It is still an open problem to extend these approaches to—or develop new ones for—the FSI case. While if full field velocity measurements are available the approach of Tiago et al.¹¹¹ allows to solve for the fluid only—implicitly considering the arterial wall kinematics from the data—in the case of FSI it is an open problem how the coupling conditions should be formulated.

10 | **STATE ESTIMATION**

10.1 | Preliminaries

Blood flows are time-dependent due to the pulsatile nature of the cardiovascular physiology. Therefore, dealing with time-dependent effects is important. The evolution of the flow dynamics essentially depends on the initial condition, X (t_0), which in the general case is unknown and has to be estimated from data. The present section reviews the literature addressing the issue of state estimation under uncertain initial conditions, again structured by the optimization method employed.

10.2 | Rigid-wall models

10.2.1 | Variational approaches

To the best of our knowledge, the only work fully estimating the initial condition in blood flows is Funke et al.,¹²¹ using variational DA and simultaneously estimating the time-varying velocity boundary conditions. A detailed discussion of their work was already presented in the context of boundary condition estimation, cf. Section 6.4.2.

10.2.2 | Kalman filter approaches

In Habibi et al.,¹⁵³ a linear Kalman filter was applied to estimate the state from synthetic 4D Flow data in aneurysm geometry. For the filter to become tractable, the system dynamics were reduced using dynamic mode decomposition¹⁵⁴ generated with errors of about 10% in the inflow velocity and viscosity. In 3D, improvements with respect to the ground truth were about 70% compared to the non-filtered case.

10.2.3 | Luenberger observer approaches

Velocity feedbacks were proposed in Funamoto et al.^{155,156} in the context of blood flows, using Doppler-like synthetic velocity measurements. The study, however, did not assess recovering the state trajectory from an uncertain initial condition but by setting up the observer with incorrect boundary conditions. It was shown that the observer could also reduce the error with respect to the reference simulation in that situation.

An important remark is that velocity observers in incompressible flows based on localized velocity measurements (e.g., slices) is that the effect of the feedback is localized around the measurement location due to the parabolic properties of the fluid system.¹⁵⁷ Therefore, effective fluid observers based on velocity measurements should be ideally based on full field measurements such as 4D Flow.

Additionally, observers in incompressible flow can only be proven to reduce the velocity error in the energy norm, hence there is no control on the pressure error. Indeed, Funamoto and Hayase¹⁵⁸ showed numerically that velocity observers may produce spurious pressures. Also, a pressure compensation method was devised where the observer's pressure was corrected a posteriori in the regions where the divergence of the feedback term is large.¹⁵⁸ In order words, this is equivalent to taking only the divergence-free part of the feedback term in order to avoid pressure perturbations. Results showed to be more accurate than the approach without any correction from Funamoto et al.¹⁵⁶ In Ii et al.,¹⁵⁹ the approach was further refined to applying then divergence-free feedback term only at inlet and outlet boundaries,

showing that the observer presents reduced perturbations from to the measurement noise with respect to the feedback applied on the whole domain. As the authors claimed, theoretical analysis that supports the theoretical results remains an open task.

10.3 | Fluid-structure interaction models

There is only one reported work on state estimation in FSI problems,¹⁶⁰ where the performance of the observers presented in Moireau et al.^{161,162} based on displacement and velocity measurements in the solid was studied theoretically and numerically in the FSI framework. It was shown that the straightforward usage of the these estimators in FSI leads to a considerably better performance of the displacement with respect to the velocity feedback, while in pure solid mechanics usually the opposite occurs. The theoretical analysis concluded that the velocity feedback does not take into account (by construction) the added mass effect in the FSI case. Hence, an improved observer was proposed by including the added mass operator in the feedback. It was also shown that to accelerate the convergence of the observer to the target trajectory for the full fluid–solid state also velocity measurements in the fluid are needed.

10.4 | Discussion and perspectives

Independent of the method, ideally both initial condition and parameters should be estimated. In the context of variational data assimilation, incorporating the initial condition seems still to be an open problem for FSI problems. The methods and theory developed by Failer¹⁶³ for the control of FSI problems can be extended to tackle the initial condition estimation.

As mentioned in Section 5.3, the full estimation of the initial condition with a Kalman filter is computationally prohibitive. An approach to partially tackle this issue is to assume that the initial condition belongs to some (linear) subspace of reduced dimension, and include those degrees of freedom as uncertain parameters.¹⁶⁴

Joint parameter and state estimation can also be performed by combining Luenberger observers and Kalman filters, as was successfully done in solid mechanics.¹⁰⁴ In Luenberger observers with perfectly known parameters, the error in the whole state in the energy norm decreases over time. However, in multi-physics problems when (partial) observations in only one of the fields are available, it may happen that the error in the fields which are not measured increases. This may negatively impact the estimation of the parameters when Luenberger and Kalman filters are combined.

Another challenge to be addressed in Luenberger observer problems in FSI is the choice of the scalar gain within \mathscr{L} , since it results from computationally expensive eigenvalue.

11 | FURTHER TOPICS

The literature reviewed in this article was limited to settings considering the full 3D/2D models and the estimation of their material properties, boundary conditions and initial conditions. In this section, we list some of the works dealing with related topics, without presenting exhaustive commentary. We believe that this may be of interest to the readers of this article.

11.1 | Inverse uncertainty quantification

Bayesian parameter estimation—as given in Section 5.1—enables uncertainty quantification (UQ) of the parameter estimates, as parameters and the corresponding model outputs are considered probabilistic distributions.

It should be first noted that parameter estimation based on Kalman filters (references^{31, 33, 43, 104, 118, 119, 123, 124, 143, 146, 151, 152} in Sections 6 to 9, and references^{34, 164–169} for GROMs in Section 11.4) inherently includes uncertainty quantification via tracking of the covariance matrix (following an assumption of Gaussian distributions) of the parameters. As follows, we will review research based on other types of solution methods for the Bayesian estimation problem.

Lassila et al.¹⁷¹ presented methods of computational and geometric order reduction which were applied to deterministic and Bayesian parameter estimation problems. For a steady FSI model of a 2D idealized stenotic vessel, employing a reduced basis, parameter distributions of the elastic wall model were inferred from measurements of the pressure drop. The Young's modulus could be reliably identified. The estimation proved very sensitive to noise. Despite the shear modulus resulting to be unidentifiable, distributions of the wall shear stress could be estimated from pressure drop data by means of Monte Carlo sampling. A second example explores optimal bypass design for a occluded femoral artery and estimates the residual flow distribution from pressure drop measurements.

Further publications study inverse UQ in geometrically reduced problems (see also Section 11.4), and shall be listed only briefly and with no claim to completeness. Considering 1D arterial networks, Arnold et al.¹⁷² studied the estimation of inflow waveforms from ex vivo pressure and vessel area measurements in a probabilistic setting using the EnKF and providing UQ for the inflow, pressure and vessel area estimates. Paun et al.^{173,174} investigated Bayesian parameter estimation and UQ using Markov Chain Monte Carlo sampling and Gaussian processes for 1D models of the pulmonary arteries in mice. Arterial wall stiffness and Windkessel boundary parameters were inferred from in vivo and synthetic data. Schiavazzi et al.¹⁷⁵ perform UQ in the context of virtual heart surgery. First using an LPM, Bayesian inversion relates uncertain clinical data on the pressure and flow split ratio with Windkessel parameter output distributions. In the second step consists in the forward propagation of the parameter uncertainty with a 3D Navier-Stokes model, where the Windkessel parameters enter as boundary conditions. Similarly, in Tran et al.,¹⁷⁶ Bayesian parameter estimation problems were solved via Monte Carlo sampling in LPMs of coronary arteries, to recover hemodynamic parameter distributions from non-invasive clinical data. These were in a separate step propagated with a full multiscale model to obtain confidence intervals of 3D results. Larson et al.¹⁷⁷ formulated a Bayesian parameter estimation and UQ framework for bifurcating arterial 1D networks. Given noisy, synthetic velocity observations, they estimated the blood viscosity, arterial stiffness, and vessel area, and identify structural defects (modeled via perturbations in the latter two parameters).

11.2 | State estimation by linear optimization

Rispoli et al.¹⁷⁸ presented a linear least squares method to denoise and increase the spatial resolution of 4D Flow data. The velocity estimate is the one minimizing the weighted sum of two quadratic terms: (a) the discrepancy between the velocity estimate and 4D Flow data and (b) a term that penalizes the Navier–Stokes model residual using a time-explicit discretization. The velocity estimate results in a weighting of a Navier–Stokes solution and the measured data. The approach may therefore be interpreted as data denoising or model correction, depending on the values of the weights.

With a similar purpose a different approach was presented by Fathi et al.¹⁷⁹ A forward Navier–Stokes simulation with boundary conditions based on the 4D Flow rates served to generate snapshots from which a reduced basis was constructed using *proper orthogonal decomposition* (POD). A velocity estimate in the POD basis was found by balancing the quadratic error with respect to 4D Flow measurements and a lasso ℓ_1 regularization promoting sparsity in the POD basis. Comparisons for in vivo 4D Flow data to other denoising methods showed a superior performance of the POD-lasso approach. However, the question of the choice of the penalization parameter of the ℓ_1 term remains open.

With the aim of extrapolating 2D measurements to the whole 3D domain using the flow physics, Galarce et al.^{180,181} proposed an approach to find the element of a linear subspace (e.g., POD basis) closest to the true 3D flow velocity solution such that the estimate matches the measurements. This match can be performed exactly or in a least squares sense. The number of elements of the POD basis was chosen such that a priori error bounds for a quantity of interest of the flow (velocity, pressure drop) were minimized. Moreover, an inequality constraint on the basis coefficients was included, avoiding penalization parameters in the optimization functional. The approach was extended in Galarce et al.¹⁸² to account for domain shape uncertainties when the domain's geometry is not easily parametrizable.

It is interesting to remark that the formulations presented in these works^{179–181} are complementary, for instance by combining the ℓ_1 regularization¹⁷⁹ with the a priori mode selection and coefficient constraints.^{180,181} Concerning the computational cost, both cases are dominated by the number of forward solutions. In Fathi et al.,¹⁷⁹ only one forward simulation was used to generate the snapshots for the POD basis because the boundary conditions were extracted from the 4D Flow data. Since in Galarce et al.^{180,181} only a smaller number of (sparse) measurements was assumed (like 2D Doppler ultrasonography images), the basis was generated by probing the flow's model parameter space (e.g., boundary conditions) leading to an important number of forward computations. However, the strategy for generating snapshots is independent on the estimation method and rather depends on the data availability.

All the works mentioned in this section have focused on rigid wall models, and therefore their extension to FSI remains to be investigated.

11.3 | Machine learning

Machine learning techniques, in particular "deep learning" algorithms using deep neural networks (DNN), have made their entry into the field of computational medicine during the last decade. Publications on deep learning in medical image analysis seem to be growing exponentially and numerous recent and heavily cited literature surveys^{183–187} have appeared, trying to keep track of the dynamic developments. Deep learning is already being employed for the classification of exams and image features, for the detection and location of organs and lesions, has become the gold standard in image segmentation, is used for registration and other purposes.¹⁸³ Related to image-based hemodynamic analysis, first proofs of concept propose deep learning-based data-driven surrogate models for aortic flows,^{188–190} which have the potential to become useful in the context of inverse hemodynamics in the future. However, research of machine learning techniques to solve inverse problems in hemodynamics remains scarce, although a growing number of publication can be expected following the general trend in other fields.

An approach to include physical models in neural networks was proposed by Raissi et al.,¹⁹¹ deviating from the purely data-driven paradigm of machine learning. Their physics-informed neural networks (PINN) enable state estimation and parameter identification. Making use of the universal function approximator property of DNNs, functions (e.g., blood velocity) are expressed as DNNs that take time and space coordinates as inputs. Automatic differentiation of DNNs allows evaluating the residual of PDEs governing the approximated functions, which is minimized during the learning process of the DNNs. Raissi et al.¹⁹² used a PINN to accurately recover the velocity and pressure fields in a 3D intracranial aneurysm from observations of a scalar tracer concentration, using synthetic data. An advantage over patient-specific simulations is that no flow boundary conditions are required. Kissas et al.¹⁹³ employed PINNs to predict the velocity and pressure wave propagation in a 1D compliant vessel model from 4D Flow velocity and wall displacement measurements. Their method avoids mesh generation and the prescription of boundary conditions. It requires large per-patient training sets, which could possibly be alleviated by transfer learning methods. Fathi et al.¹⁹⁴ explored the reconstruction of a denoised, high-resolution flow velocity from 4D Flow data using a physics-informed DNN following Raissi et al.,¹⁹¹ adapted to the specific properties of 4D Flow data acquisition. An in vitro validation using high-resolution PIV measurements as reference showed a qualitative improvement compared to the input 4D Flow data, but no significant improvement in the error metrics.

In addition, the PINN framework allows parameter estimation,¹⁹¹ but this feature has not yet been applied to problems in hemodynamics. Regazzoni et al.¹⁹⁵ combine data assimilation of cardiovascular flow (a two-stage Windkessel model) with machine learning to identify an evolution law of unobservable, slowly evolving parameters (Windkessel resistances). A neural network is used to express the right hand side of the evolution ODE and calibrated by means of adjoint-based solution of a minimization problem. The framework is applied to a data-driven model of hypertension development in a synthetic, idealized setting and provides accurate results.

11.4 | Calibration of stand-alone lumped-parameter and 1D models

When flow rates are available at each of the outlets of a 3D geometry, and some information is given on the pressure wave—i.e., either from invasive measurements or an assumed wave form calibrated using non-invasive diastolic/ systolic cuff measurements—the LPM parameters can be estimated independently for each of the outlet, as, for example, done by Romarowski et al.¹⁹⁶

In cases where data such as flow rates is distributed over the arterial tree, data assimilation techniques for the calibration of 1D networks or LPMs of (parts of) the cardiovascular system have been applied. Here, we briefly summarize the literature, with no claim to completeness.

Single or coupled multicompartment LPMs—governed by ODEs independent of spatial coordinates—have been calibrated with measured flow rates and pressure data traditionally by means of nonlinear least-squares optimization, following the early works^{197–199} of the 1980s. Since then, similar approaches, successively refined, have been used in LPMs of the cardiovascular system.^{200–204}

Schiavazzi et al.^{175,205} and Tran et al.¹⁷⁶ used Bayesian estimation approaches (Markov Chain Monte Carlo) for the identification of parameters in complex circulation models, while Pant et al.^{165,166} performed sequential data assimilation in form of the UKF.

Parameter estimation in 1D GROMs, modeling the cardiovascular system as a network of compliant pipes, was undertaken by, for example, Cousins and Gremaud²⁰⁶ using least-squares optimization and special structured-tree

boundary conditions²⁰⁷ and by Blanco et al.²⁰⁸ by means of a root-finding problem. Ventre et al.²⁰⁹ and Carson et al.²¹⁰ calibrate staged 0D-1D approaches via least-squares optimization and root-finding, respectively. More flexibility regarding parameters and data is offered by sequential data assimilation, for example, by means of the EnKF^{167,168} or the ROUKF.^{34,169,170} Bayesian parameter estimation was conducted in several publications,^{172–174,177} briefly discussed in Section 11.1.

11.5 | Shape estimation of vessels and obstacles

As indicated in Section 9, velocity measurements allow to estimate the shape of obstacles within a vessel. This can be an interesting approach for instance for cardiac valve or stents, which are hard to visualize in non-ionizing images like MRI.

This problem can be tackled by shape optimization approaches (see, e.g., Quarteroni and Rozza,²¹¹ Agoshkov et al.²¹² and Manzoni and Ponti²¹³). It is important to notice that shape optimization, when the domain is parametrizable, can be accelerated using model reduction techniques, as done in Lassila et al.¹⁷¹ for the case of design of bypass grafts.

A simpler approach was presented in Aguayo et al.²¹⁴ and applied to cardiac valves by means of estimating a resistance function approximating the obstacle in the stationary Navier–Stokes equations from velocity measurements.

ACKNOWLEDGEMENTS

This work was partially funded by the European Research Council (ERC) under the European Union's Horizon 2020 research and innovation programme (grant 852544—CardioZoom). We also acknowledge the authors of some of the references cited here who have dedicated time to discuss and clarify aspects of their work.

DATA AVAILABILITY STATEMENT

Data sharing is not applicable to this article as no new data were created or analyzed in this study.

ORCID

Cristóbal Bertoglio D https://orcid.org/0000-0001-5049-1707

REFERENCES

- 1. WHO. Global Heath Estimates 2016: Deaths by Cause, Age, Sex, by Country and by Region, 2000–2016. Tech. rep. World Health Organization;: 2018.
- 2. Wilkins E, Wilson L, Wickramasinghe K, et al. (2017). European Cardiovascular Disease Statistics 2017. Tech. rep., European Heart Network.
- Baumgartner H, Hung J, Bermejo J, Chambers J.B., Evangelista A., Griffin B.P., Iung B., Otto C.M., Pellikka P.A., Quiñones M., American Society of Echocardiography., European Association of Echocardiography. Echocardiographic assessment of valve stenosis: EAE/-ASE recommendations for clinical practice. J Am Soc Echocardiogr 2009–01; 22(1): 1–23. doi: 10.1016/j.echo.2008.11.029
- 4. Baumgartner H, Bonhoeffer P, De Groot NM, et al. ESC guidelines for the Management of Grown-up Congenital Heart Disease (new version 2010) the task force on the Management of Grown-up Congenital Heart Disease of the European Society of Cardiology (ESC). *Eur Heart J.* 2010;31(23):2915-2957.
- 5. Jenkins NP, Ward C. Coarctation of the aorta: natural history and outcome after surgical treatment. Qjm. 1999;92(7):365-371.
- Cioffi G, Faggiano P, Vizzardi E, et al. Prognostic effect of inappropriately high left ventricular mass in asymptomatic severe aortic stenosis. *Heart*. 2011;97(4):301-307.
- 7. Kilner PJ. Imaging congenital heart disease in adults. Br J Radiol. 2011;84:S258-S268. doi:10.1259/bjr/74240815
- Cavalcante J, Lima J, Redheuil A, Al-Mallah M. Aortic stiffness: current understanding and future directions. J Am Coll Cardiol. 2011; 57(14):1511-1522.
- 9. Oparil S, Acelajado MC, Bakris GL, et al. Hypertension. Nat Rev Dis Primers. 2018;4:18014 EP.
- 10. Li X, Zhao G, Zhang J, Duan Z, Xin S. Prevalence and trends of the abdominal aortic aneurysms epidemic in general population-a meta-analysis. *PLoS One*. 2013;8(12):e81260.
- 11. Farzaneh S, Trabelsi O, Avril S. Inverse identification of local stiffness across ascending thoracic aortic aneurysms. *Biomech Model Mechanobiol.* 2018;18:1-17.
- 12. Taylor CA, Steinman DA. Image-based modeling of blood flow and vessel wall dynamics: applications, methods and future directions. *Ann Biomed Eng.* 2010;38(3):1188-1203.

MILEY-

- 13. D'Elia M, Mirabella L, Passerini T, et al. Applications of Variational Data Assimilation in Computational Hemodynamics. Springer; 2012:363-394.
- Veneziani A, Vergara C. Inverse problems in cardiovascular mathematics: toward patient-specific data assimilation and optimization. Int J Numer Methods Biomed Eng. 2013;29(7):723-725. doi:10.1002/cnm.2566
- 15. Marsden AL. Optimization in cardiovascular modeling. Annu Rev Fluid Mech. 2014;46:519-546.
- 16. Quarteroni A, Manzoni A, Vergara C. The cardiovascular system: mathematical modelling, numerical algorithms and clinical applications. *Acta Numer.* 2017;26:365-590.
- 17. Quarteroni A, Dedé L, Manzoni A, Vergara C, others. *Mathematical Modelling of the Human Cardiovascular System: Data, Numerical Approximation, Clinical Applications.* 33. Cambridge University Press. 2019.
- Baskurt OK, Meiselman HJ. Blood rheology and hemodynamics. Sem Thromb Hemost 2003–10; 29(5): 435–450.doi:10.1055/s-2003-44551
- 19. Baskurt OK, Hardeman MR, Rampling MW., eds. Handbook of Hemorheology and Hemodynamics. Vol. 69 of biomedical and Health Research. IOS Press (1st ed). 2007. OCLC: ocn183145995.
- 20. Bird RB. Dynamics of Polymeric Liquids. 2nd ed. Wiley; 1987.
- Gijsen F, Vosse V. DF, Janssen J. The influence of the non-Newtonian properties of blood on the flow in large arteries: steady flow in a carotid bifurcation model. J Biomech 1999/2000 6; 32(6): 601–608. doi: 10.1016/S0021-9290(99)00015-9
- 22. Cho YI, Kensey KR. Effects of the non-Newtonian viscosity of blood on flows in a diseased arterial vessel. Part 1: steady flows. *Bio-rheology* 1991;28(3-4):241-262. doi: 10.3233/BIR-1991-283-415
- Johnston BM, Johnston PR, Corney S, Kilpatrick D. Non-Newtonian blood flow in human right coronary arteries: steady state simulations. J Biomech 2004;37(5):709–720. doi: 10.1016/j.jbiomech.2003.09.016
- 24. Tada S, Tarbell JM. Hemodynamics in physio- and pathological vessels. In: Tanishita K, Yamamoto K, eds. Vascular engineering: new Prospects of Vascular Medicine and Biology with a Multidiscipline Approach. Springer; 2016:69-98.
- 25. Bertoglio C, Caiazzo A, Bazilevs Y, et al. Benchmark problems for numerical treatment of backflow at open boundaries. *Int J Numer Methods Biomed Eng.* 2018;34(2):e2918.
- 26. Nichols W, O'Rourke M, Vlachopoulos C. McDonald's blood flow in arteries, sixth edition: theoretical, experimental and clinical principles. 6th ed. CRC Press; 2011 OCLC: 1027501178.
- Pacheco D, Steinbach O. A continuous finite element framework for the pressure Poisson equation allowing non-Newtonian and compressible flow behavior. Int J Numer Meth Fluids. 2020;93:1-11. doi:10.1002/fld.4936
- 28. Holzapfel GA, Gasser TC, Ogden RW. A new constitutive framework for Arterial Wall mechanics and a comparative study of material models. *J Elast Phys Sci Sol.* 2000;61(1–3):1-48.
- 29. Holzapfel GA, Ogden RW. Constitutive modelling of arteries. Proc R Soc A: Math, Phys Eng Sci. 2010;466(2118):1551-1597.
- Holzapfel GA, Ogden RW, Sherifova S. On fibre dispersion modelling of soft biological tissues: A review. Proc R Soc A. 2019;475(2224): 20180736.
- Moireau P, Xiao N, Astorino M, et al. External tissue support and fluid-structure simulation in blood flows. Biomech Model Mechanobiol. 2012;11(1):1-18.
- 32. Peiró J, Veneziani A. Reduced models of the cardiovascular system. In: Formaggia L, Quarteroni A, Veneziani A, eds. Cardiovascular Mathematics: Modeling and Simulation of the Circulatory System. MS&A; 2009:347-394.
- Pant S, Fabrèges B, Gerbeau JF, Vignon-Clementel IE. A methodological paradigm for patient-specific multi-scale CFD simulations: from clinical measurements to parameter estimates for individual analysis. *Int J Numer Methods Biomed Eng* 2014;30(12):1614–1648. doi: 10.1002/cnm.2692
- 34. Lombardi D. Inverse problems in 1D hemodynamics on systemic networks: a sequential approach. *Int J Numer Methods Biomed Eng.* 2014;30(2):160-179.
- 35. Westerhof N, Lankhaar JW, Westerhof BE. The arterial Windkessel. *Med Biol Eng Comput.* 2009;47(2):131-141.
- Formaggia L, Quarteroni A, Veneziani A. Multiscale models of the vascular system. In: Formaggia L, Quarteroni A, Veneziani A, eds. Cardiovascular Mathematics: Modeling and Simulation of the Circulatory System. Springer; 2009:395-446.
- Quarteroni A, Veneziani A, Vergara C. Geometric multiscale modeling of the cardiovascular system, between theory and practice. Comput Methods Appl Mech Eng. 2016;302:193-252.
- 38. Fleischmann D, Chin AS, Molvin L, Wang J, Hallett R. Computed tomography angiography: a review and technical update. *Radiol Clin*. 2016;54(1):1-12.
- 39. Mayo JR, Leipsic JA. Radiation dose in cardiac CT. Am J Roentgenol. 2009;192(3):646-653.
- Nakanishi R, Rajani R, Ishikawa Y, Ishii T, Berman DS. Myocardial bridging on coronary CTA: an innocent bystander or a culprit in myocardial infarction?. J Cardiovasc Comput Tomogr 2012; 6(1): 3–13. doi:10.1016/j.jcct.2011.10.015
- 41. Nolte D, Urbina J, Sotelo J, et al. Validation of 4D-flow-based pressure difference estimators. Working Paper or Preprint; 2021.
- 42. McRobbie DW, Moore EA, Graves MJ, Prince MR. MRI from Picture to Proton. 3rd ed. Cambridge University Press; 2017.
- Bertoglio C, Barber D, Gaddum N, et al. Identification of artery wall stiffness: in vitro validation and in vivo results of a data assimilation procedure applied to a 3D fluid-structure interaction model. J Biomech. 2014;47(5):1027-1034. doi:10.1016/j.jbiomech.2013.12.029
- 44. Smith NB, Webb A. Introduction to Medical Imaging: Physics, Engineering and Clinical Applications. 1st ed. Cambridge University Press; 2010.
- 45. Lee W. General principles of carotid Doppler ultrasonography. Ultrasonography. 2013;33(1):11-17. doi:10.14366/usg.13018

- Taylor CA, Draney MT. Experimental and computational methods in cardiovascular fluid mechanics. *Annu Rev Fluid Mech* 2004;36(1): 197–231. doi: 10.1146/annurev.fluid.36.050802.121944
- 47. Brown RW, Cheng YCN, Haacke EM, Thompson MR, Venkatesan R. Magnetic Resonance Imaging. *Physical Principles and Sequence Design*. 2nd ed. John Wiley & Sons, Inc; 2014.
- 48. Hom JJ, Ordovas K, Reddy GP. Velocity-encoded cine MR imaging in aortic Coarctation: functional assessment of hemodynamic events. *Radiographics*. 2008;28(2):407-416.
- 49. Markl M, Frydrychowicz A, Kozerke S, Hope M, Wieben O. 4D flow MRI. J Magn Reson Imag. 2012;36(5):1015-1036.
- 50. Dyverfeldt P, Bissell M, Barker AJ, et al. 4D flow cardiovascular magnetic resonance consensus statement. *J Cardiovasc Magn Reson.* 2015;17(1):1-19.
- Neuhaus E, Weiss K, Bastkowski R, Koopmann J, Maintz D, Giese D. Accelerated aortic 4D flow cardiovascular magnetic resonance using compressed sensing: applicability, validation and clinical integration. J Cardiovasc Magn Reson. 2019;21(1):65.
- 52. Ma LE, Markl M, Chow K, et al. Aortic 4D flow MRI in 2 minutes using compressed sensing, respiratory controlled adaptive k-space reordering, and inline reconstruction. *Magn Reson Med.* 2019;81(6):3675-3690.
- 53. Peper ES, Gottwald LM, Zhang Q, et al. Highly accelerated 4D flow cardiovascular magnetic resonance using a pseudo-spiral Cartesian acquisition and compressed sensing reconstruction for carotid flow and wall shear stress. *J Cardiovasc Magn Reson.* 2020;22(1):1-15.
- Dyverfeldt P, Sigfridsson A, Kvitting JPE, Ebbers T. Quantification of Intravoxel velocity standard deviation and turbulence intensity by generalizing phase-contrast MRI. *Magn Reson Med* 2006;56(4):850–858. doi: 10.1002/mrm.21022
- 55. Ha H, Kim GB, Kweon J, Huh HK, Lee SJ, Koo HJ, Kang JW, Lim TH, Kim DH, Kim YH, Kim N, Yang DH Turbulent kinetic energy measurement using phase contrast MRI for estimating the post-stenotic pressure drop: *In Vitro Validation and Clinical Application PLOS ONE* 2016;11(3):e0151540. doi: 10.1371/journal.pone.0151540
- 56. Ha H, Kvitting JPE, Dyverfeldt P, Ebbers T. 4D flow MRI quantification of blood flow patterns, turbulence and pressure drop in normal and stenotic prosthetic heart valves. *Magn Reson Imaging*. 2019;55:118-127.
- 57. Walheim J, Dillinger H, Kozerke S. Multipoint 5D flow cardiovascular magnetic resonance-accelerated cardiac-and respiratory-motion resolved mapping of mean and turbulent velocities. *J Cardiovasc Magn Reson.* 2019;21(1):1-13.
- 58. Pope SB. Turbulent Flows. Cambridge. 1st ed. Cambridge University Press; 2000.
- 59. Kumar R, Dubey P, Zafer A, Kumar A, Yadav S. Past, present and future of blood pressure measuring instruments and their calibration. *Measurement*. 2021;172:108845.
- 60. Babbs CF. Oscillometric measurement of systolic and diastolic blood pressures validated in a physiologic mathematical model. *Biomed Eng Online*. 2012;11(1):1-22.
- 61. Drzewiecki GM, Melbin J, Noordergraaf A. Arterial tonometry: review and analysis. J Biomech. 1983;16(2):141-152.
- 62. O'Rourke MF. Carotid artery tonometry: pros and cons. Am J Hypertens. 2016;29(3):296-298.
- 63. Wyman RM, Safian RD, Portway V, Skillman JJ, McKAY RG, Baim DS. Current complications of diagnostic and therapeutic cardiac catheterization. *J Am Coll Cardiol*. 1988;12(6):1400-1406.
- 64. Vitiello R, McCrindle BW, Nykanen D, Freedom RM, Benson LN. Complications associated with pediatric cardiac catheterization. J Am Coll Cardiol. 1998;32(5):1433-1440.
- 65. Omran H, Schmidt H, Hackenbroch M, et al. Silent and apparent cerebral embolism after retrograde catheterisation of the aortic valve in Valvular stenosis: a prospective. *Random Study Lancet*. 2003;361(9365):1241-1246.
- 66. Brüning J, Hellmeier F, Yevtushenko P, Kühne T, Goubergrits L. Uncertainty quantification for non-invasive assessment of pressure drop across a coarctation of the aorta using CFD. *Cardiovasc Eng Technol.* 2018;9(4):582-596.
- 67. Vecchi DA, Clough RE, Gaddum NR, et al. Catheter-induced errors in pressure measurements in vessels: an in-vitro and numerical study. *IEEE Trans Biomed Eng.* 2014;61(6):1844-1850.
- 68. Baumgartner H, Stefenelli T, Niederberger J, Schima H, Maurer G. "Overestimation" of catheter gradients by Doppler ultrasound in patients with aortic stenosis: a predictable manifestation of pressure recovery. *J Am Coll Cardiol*. 1999;33(6):1655-1661.
- 69. Firstenberg MS, Smedira NG, Greenberg NL, et al. Relationship between early diastolic intraventricular pressure gradients, an index of elastic recoil, and improvements in systolic and diastolic function. *Circulation*. 2001;104(suppl_1):I-330.
- 70. Marlevi D, Balmus M, Hessenthaler A, et al. Non-invasive estimation of relative pressure for intracardiac flows using virtual workenergy. *Med Image Anal*. 2021;68:101948.
- 71. Yang GZ, Kilner PJ, Wood NB, Underwood SR, Firmin DN. Computation of flow pressure fields from magnetic resonance velocity mapping. *Magn Reson Med.* 1996;36(4):520-526.
- 72. Ebbers T, Wigstrom L, Bolger AF, Wranne B, Karlsson M. Noninvasive measurement of time-varying three-dimensional relative pressure fields within the human heart. *J Biomech Eng.* 2002;124(3):288-293.
- 73. Krittian SBS, Lamata P, Michler C, Nordsletten DA, Bock J, Bradley CP, Pitcher A, Kilner PJ, Markl M, Smith NP A finite-element approach to the direct computation of relative cardiovascular pressure from time-resolved MR velocity data. *Med Image Anal* 2012; 16(5):1029–1037. doi: 10.1016/j.media.2012.04.003
- 74. Cayco ME, Nicolaides RA. Finite element technique for optimal pressure recovery from stream function formulation of viscous flows. *Math Comput.* 1986;46(174):371-377.
- Švihlová H, Hron J, Málek J, Rajagopal KR, Rajagopal K. Determination of pressure data from velocity data with a view toward its application in cardiovascular mechanics. Part 1. Theoretical considerations. *Int J Eng Sci.* 2016;105:108-127. doi:10.1016/j.ijengsci.2015. 11.002

- 76. Garay J, Mella H, Sotelo J, et al. A new mathematical model for verifying the Navier-stokes compatibility of 4D flow MRI data. Working paper or preprint; 2021.
- 77. Bertoglio C, Nuñez R, Galarce F, Nordsletten D, Osses A. Relative pressure estimation from velocity measurements in blood flows: state-of-the-art and new approaches. *IJNMBE*. 2018;34(2):e2925. doi:10.1002/cnm.2925
- 78. Araya R, Bertoglio C, Cárcamo C, Nolte D, Uribe S. Error analysis of pressure reconstruction from discrete velocities. HAL Preprint HAL-03186241-v2. 2021.
- 79. Pacheco DR. On the numerical treatment of viscous and convective effects in relative pressure reconstruction methods. *Int J Numer Methods Biomed Eng.* 2021;38:e3562.
- 80. Donati F, Figueroa CA, Smith NP, Lamata P, Nordsletten DA. Non-invasive pressure difference estimation from PC-MRI using the work-energy equation. *Med Image Anal.* 2015;26(1):159-172.
- Donati F, Myerson S, Bissell MM, et al. Beyond Bernoulli: improving the accuracy and precision of noninvasive estimation of peak pressure drops. *Circul: Cardiovasc Imag.* 2017;10(1):e005207. doi:10.1161/CIRCIMAGING.116.005207
- Marlevi D, Ruijsink B, Balmus M, Dillon-Murphy D, Fovargue D, Pushparajah K, Bertoglio C, Colarieti-Tosti M, Larsson M, Lamata P, Figueroa CA, Razavi R, Nordsletten DA Estimation of cardiovascular relative pressure using virtual work-energy. *Sci Rep* 2019;9(1): 1375. doi: 10.1038/s41598-018-37714-0
- Marlevi D, Schollenberger J, Aristova M, et al. Noninvasive quantification of cerebrovascular pressure changes using 4D flow MRI. Magn Reson Med. 2021;86:3096-3110.
- 84. Marlevi D, Mariscal-Harana J, Burris NS, et al. Altered aortic hemodynamics and relative pressure in patients with dilated cardiomyopathy. *J Cardiovasc Transl Res.* 2021 [Online ahead of print]. doi:10.1007/s12265-021-10181-1
- 85. Haraldsson H, Kefayati S, Ahn S, et al. Assessment of Reynolds stress components and turbulent pressure loss using 4D flow MRI with extended motion encoding. *Magn Reson Med.* 2018;79(4):1962-1971.
- 86. Marlevi D, Ha H, Dillon-Murphy D, et al. Non-invasive estimation of relative pressure in turbulent flow using virtual work-energy. *Med Image Anal.* 2020;60:101627.
- 87. Le Dimet FX, Talagrand O. Variational algorithms for analysis and assimilation of meteorological observations: theoretical aspects. *Tellus A: Dyn Meteorol Oceanogr.* 1986;38(2):97-110.
- 88. Talagrand O. Assimilation of observations, an introduction. J Meteorol Soci Japan Ser. 1997;II(75):191-209.
- 89. Talagrand O. Variational assimilation. In: Lahoz W, Khattatov B, Menard R, eds. Data Assimilation. Springer; 2010:41-67.
- Farrell PE, Ham DA, Funke SW, Rognes ME. Automated derivation of the Adjoint of high-level transient finite element programs. SIAM J Scient Comput. 2013;35(4):C369-C393.
- 91. Griewank A, Walther A. Algorithm 799: revolve: an implementation of Checkpointing for the reverse or Adjoint mode of computational differentiation. *ACM Trans Math Softw.* 2000;26(1):19-45.
- 92. Gunzburger MD. Perspectives in Flow Control and Optimization. Society for Industrial and Applied Mathematics. first ed; 2003.
- 93. Naumann U. The art of differentiating computer programs: an introduction to algorithmic differentiation. *Software, Environments, and ToolsPhiladelphia: Society for Industrial and Applied Mathematics.* SIAM; 2012.
- 94. Kalman R, Bucy R. New results in linear filtering and prediction theory. Trans ASME J Basic Eng. 1961;83:95-108.
- 95. Asch M, Bocquet M, Nodet M. Data assimilation: methods. In Algorithms, and Applications. SIAM. 2016.
- 96. Evensen G. Data Assimilation: the Ensemble Kalman Filter. 2nd ed. Berlin Heidelberg; 2009.
- Hunt BR, Kostelich EJ, Szunyogh I. Efficient data assimilation for spatiotemporal chaos: A local ensemble transform Kalman filter. Phys D: Nonlinear Phenom 2007; 230(1–2):112–126. doi:10.1016/j.physd.2006.11.008, 112
- 98. Julier SJ, Uhlmann JK. New extension of the Kalman filter to nonlinear systems. In: Kadar I, ed. Signal Processing, Sensor Fusion, and Target Recognition VI. 3068. SPIE, International Society for Optics and Photonics; 1997:182-193.
- 99. Kalnay E, Li H, Miyoshi T, Yang SC, Ballabrera-Poy J. 4-D-Var or ensemble Kalman filter?. *Tellus A: Dyn Meteorol Oceanogr* 2007-01-01; 59(5): 758–773. doi: 10.1111/j.1600-0870.2007.00261.x
- Moireau P, Chapelle D. Reduced-order unscented Kalman filtering with application to parameter identification in large-dimensional systems. ESAIM: Control, Optim Calc Var. 2011;17(2):380-405.
- Pham DT, Verron J, Gourdeau L. Filtres de Kalman singuliers évolutifs pour l'assimilation de données en océanographie. C. R. Acad. Sci. Ser. IIA. 1998;326(4):255-260. doi:10.1016/S1251-8050(97)86815-2
- Hoteit I, Pham DT, Blum J. A simplified reduced order Kalman filtering and application to altimetric data assimilation in tropical Pacific. J Mar Syst. 2002;36(1/2):101-127. doi:10.1016/S0924-7963(02)00129-X
- 103. Luenberger DG. An introduction to observers. IEEE Trans Autom Control. 1971;16:596-602.
- 104. Chapelle D, Moireau P, Le Tallec P. Robust filtering for joint state-parameter estimation in distributed mechanical systems. DCDS-A 2009; 23(1/2): 65–84. A special issue Dedicated to Ta-Tsien Li on the Occasion of his 70th Birthday.
- Formaggia L, Veneziani A, Vergara C. A new approach to numerical solution of defective boundary value problems in incompressible fluid dynamics. SIAM J Numer Anal. 2008;46(6):2769-2794.
- Formaggia L, Veneziani A, Vergara C. Flow rate boundary problems for an incompressible fluid in deformable domains: formulations and solution methods. *Comput Methods Appl Mech Eng.* 2010;199(9–12):677-688.
- D'Elia M, Veneziani A. Methods for assimilating blood velocity measures in hemodynamics simulations: preliminary results. Proc Comput Sci. 2010;1(1):1231-1239.

- D'Elia M, Perego M, Veneziani A. A Variational data assimilation procedure for the incompressible Navier-stokes equations in hemodynamics. J Scient Comput 2012;52(2):340–359. doi: 10.1007/s10915-011-9547-6
- 109. Adib MAHM, Ii S, Watanabe Y, Wada S. Minimizing the blood velocity differences between phase-contrast magnetic resonance imaging and computational fluid dynamics simulation in cerebral arteries and aneurysms. *Med Biol Eng Comput.* 2017;55(9):1605-1619.
- 110. Zainib Z, Ballarin F, Fremes S, Triverio P, Jiménez-Juan L, Rozzã G. Reduced order methods for parametric optimal flow control in coronary bypass grafts, toward patient-specific data a ssimilation. *Int J Numer Methods Biomed Eng.* 2020;37:e3367.
- 111. Tiago J, Gambaruto A, Sequeira A. Patient-specific blood flow simulations: setting Dirichlet boundary conditions for minimal error with respect to measured data. *Math Model Nat Phenom*. 2014;9(6):98-116.
- 112. Guerra T, Tiago J, Sequeira A. Optimal control in blood flow simulations. Int J Non-Linear Mech. 2014;64:57-69.
- Tiago J, Guerra T, Sequeira A. A velocity tracking approach for the data assimilation problem in blood flow simulations. Int J Numer Methods Biomed Eng. 2017;33(10):e2856. doi:10.1002/cnm.2856
- 114. Guerra T, Catarino C, Mestre T, Santos S, Tiago J, Sequeira A. A data assimilation approach for non-Newtonian blood flow simulations in 3D geometries. *Appl Math Comput* 2018;321:176–194. doi: 10.1016/j.amc.2017.10.029
- 115. Reyes D. IJC, Yousept I. Regularized state-constrained boundary optimal control of the Navier–stokes equations. J Math Anal Appl. 2009;356(1):257-279.
- 116. Koltukluoğlu TS, Blanco PJ. Boundary control in computational Haemodynamics. J Fluid Mech 2018;847:329–364. doi: 10.1017/jfm. 2018.329
- 117. Koltukluoğlu TS. Fourier spectral dynamic data assimilation: interlacing CFD with 4D flow MRI. International Conference on Medical Image Computing and Computer-Assisted Intervention MICCAI 2019. Springer; 2019:741-749.
- 118. Gaidzik F, Stucht D, Roloff C, Speck O, Thévenin D, Janiga G. Transient flow prediction in an idealized aneurysm geometry using data assimilation. *Comput Biol Med.* 2019;115:103507.
- 119. Gaidzik F, Pathiraja S, Saalfeld S, et al. Hemodynamic data assimilation in a subject-specific circle of Willis geometry. *Clin Neuroradiol.* 2020;31:1-9.
- 120. Töger J, Zahr MJ, Aristokleous N, Markenroth Bloch K, Carlsson M, Persson PO. Blood flow imaging by optimal matching of computational fluid dynamics to 4D-flow data. *Magn Reson Med.* 2020;84:2231-2245.
- 121. Funke SW, Nordaas M, Evju Ø, Alnæs MS, Mardal KA. Variational data assimilation for transient blood flow simulations: cerebral aneurysms as an illustrative example. *Int J Numer Methods Biomed Eng.* 2019;35(1):e3152.
- 122. Fevola E, Ballarin F, Jiménez-Juan L, Fremes S., Grivet-Talocia S., Rozza G., Triverio P. An optimal control approach to determine resistance-type boundary conditions from in-vivo data for cardiovascular simulations. *Int J Numer Methods Biomed Eng* 2021; 37(10): e3516. doi:10.1002/cnm.3516
- 123. Arthurs CJ, Xiao N, Moireau P, Schaeffter T, Figueroa CA. A flexible framework for sequential estimation of model parameters in computational hemodynamics. *Adv Model Simul Eng Sci.* 2020;7(1):1-37.
- 124. Garay J, Nolte D, Löcke M, Bertoglio C. Parameter estimation in fluid flow models from aliased velocity measurements. Preprint HAL HAL-03479421; 2021.
- 125. Perdikaris P, Karniadakis GE. Model inversion via multi-Fidelity Bayesian optimization: A new paradigm for parameter estimation in Haemodynamics, and beyond. *J R Soc Interface*. 2016;13(118):20151107.
- 126. Spilker RL, Taylor CA. Tuning multidomain hemodynamic simulations to match physiological measurements. *Ann Biomed Eng.* 2010; 38(8):2635-2648.
- Ismail M, Gee MW, Wall WA. CFD challenge: hemodynamic simulation of a patient-specific aortic Coarctation model with Adjointbased calibrated Windkessel elements. *Third International Workshop, STACOM 2012.* Springer; 2013:44-52.
- 128. Ismail M, Wall WA, Gee MW. Adjoint-based inverse analysis of Windkessel parameters for patient-specific vascular models. *J Comput Phys.* 2013;244:113-130.
- 129. Itu L, Sharma P, Passerini T, Kamen A, Suciu C, Comaniciu D. A parameter estimation framework for patient-specific hemodynamic computations. *J Comput Phys.* 2015;281:316-333.
- Itu L, Sharma P, Suciu C, Moldoveanu F, Comaniciu D. Personalized blood flow computations: A hierarchical parameter estimation framework for tuning boundary conditions: hierarchical personalization of blood flow computations. *Int J Numer Methods Biomed Eng.* 2017;33(3):e02803. doi:10.1002/cnm.2803
- 131. Troianowski G, Taylor CA, Feinstein JA, Vignon-Clementel IE. Three-dimensional simulations in Glenn patients: clinically based boundary conditions, hemodynamic results and sensitivity to input data. J Biomech Eng. 2011;133(11):111006. doi:10.1115/1.4005377
- Spilker RL, Feinstein JA, Parker DW, Reddy VM, Taylor CA. Morphometry-based impedance boundary conditions for patient-specific modeling of blood flow in pulmonary arteries. Ann Biomed Eng. 2007;35(4):546-559. doi:10.1007/s10439-006-9240-3
- 133. Baretta A, Corsini C, Yang W, et al. Virtual surgeries in patients with congenital heart disease: A multi-scale modelling test case. *Philosoph Trans R Soc A: Math, Phys Eng Sci.* 2011;369:4316-4330.
- 134. Arbia G, Corsini C, Baker C, Pennati G, Hsia TY, Vignon-Clementel IE. Pulmonary hemodynamics simulations before stage 2 single ventricle surgery: patient-specific parameter identification and clinical data assessment. *Cardiovasc Eng Technol.* 2015;6(3):268-280. doi: 10.1007/s13239-015-0212-3
- 135. Xiao N, Alastruey J, Figueroa CA. A systematic comparison between 1-D and 3-D hemodynamics in compliant arterial models. *Int J Numer Methods Biomed Eng.* 2014;30(2):204-231. doi:10.1002/cnm.2598

- 136. Gerbeau JF, Vidrascu M. A quasi-Newton algorithm based on a reduced model for fluid-structure interaction problems in blood flows. *ESAIM. Math Model Numer Anal.* 2003;37(4):631-647.
- 137. Sommer G, Holzapfel G. 3D constitutive modeling of the biaxial mechanical response of intact and layer-dissected human carotid arteries. J Mech Behav Biomed Mater. 2012;5(1):116-128.
- 138. Gee MW, Förster C, Wall W. A computational strategy for prestressing patient-specific biomechanical problems under finite deformation. *Int J Numer Methods Biomed Eng.* 2010;26(1):52-72.
- 139. Weisbecker H, Pierce DM, Holzapfel GA. A generalized prestressing algorithm for finite element simulations of preloaded geometries with application to the aorta. *Int J Numer Methods Biomed Eng.* 2014;30(9):857-872.
- 140. Rausch MK, Genet M, Humphrey JD. An augmented iterative method for identifying a stress-free reference configuration in imagebased biomechanical modeling. *J Biomech.* 2017;58:227-231.
- 141. Liang L, Liu M, Martin C, Sun W. A machine learning approach as a surrogate of finite element analysis–based inverse method to estimate the zero-pressure geometry of human thoracic aorta. *Int J Numer Methods Biomed Eng.* 2018;34(8):e3103.
- 142. Peirlinck M, De Beule M, Segers P, Rebelo N. A modular inverse elastostatics approach to resolve the pressure-induced stress state for in vivo imaging based cardiovascular modeling. *J Mech Behav Biomed Mater.* 2018;85:124-133.
- 143. Bertoglio C, Moireau P, Gerbeau JF. Sequential parameter estimation for fluid-structure problems: application to hemodynamics. *Int J Numer Methods Biomed Eng.* 2012;28(4):434-455.
- 144. Perego M, Veneziani A, Vergara C. A variational approach for estimating the compliance of the cardiovascular tissue: an inverse fluidstructure interaction problem. *SIAM J Sc Comp.* 2011;33(3):1181-1211.
- 145. Bertagna L, Veneziani A. A model reduction approach for the variational estimation of vascular compliance by solving an inverse fluid-structure interaction problem. *Inverse Probl.* 2014;30(5):055006. doi:10.1088/0266-5611/30/5/055006
- 146. Stalhand J. Determination of human arterial wall parameters from clinical data. Biomech Model Mechanobiol. 2009;8(2):141-148.
- 147. Moireau P, Bertoglio C, Xiao N, et al. Sequential identification of boundary support parameters in a fluid-structure vascular model using patient image data. *Biomech Model Mechanobiol*. 2013;12(3):475-496. doi:10.1007/s10237-012-0418-3
- 148. Pozzi S, Domanin M, Forzenigo L, et al. A surrogate model for plaque modeling in carotids based on Robin conditions calibrated by cine MRI data. *Int J Numer Methods Biomed Eng.* 2021;37:e3447.
- 149. Moore JA, Steinman DA, Holdsworth DW, Ethier CR. Accuracy of computational hemodynamics in complex arterial geometries reconstructed from magnetic resonance imaging. *Ann Biomed Eng.* 1999;27(1):32-41. doi:10.1114/1.163
- 150. Perinajová R, Juffermans JF, Westenberg JJ, et al. Geometrically induced wall shear stress variability in CFD-MRI coupled simulations of blood flow in the thoracic aortas. *Comput Biol Med.* 2021;104385.
- 151. Minakowski P, Richter T. Finite element error estimates on geometrically perturbed domains. *J Scient Comput.* 2020;84(2):30. doi:10. 1007/s10915-020-01285-y
- 152. Nolte D, Bertoglio C. Reducing the impact of geometric errors in flow computations using velocity measurements. *Int J Numer Methods Biomed Eng.* 2019;35:e3203. doi:10.1002/cnm.3203
- 153. Habibi M, D'Souza RM, Dawson ST, Arzani A. Integrating multi-fidelity blood flow data with reduced-order data assimilation. *Comput Biol Med.* 2021;104566.
- 154. Habibi M, Dawson S, Arzani A. Data-driven pulsatile blood flow physics with dynamic mode decomposition. Fluids. 2020;5(3):111.
- 155. Funamoto K, Hayase T, Saijo Y, Yambe T. Numerical experiment for ultrasonic-measurement-integrated simulation of threedimensional unsteady blood flow. *Ann Biomed Eng.* 2008;36(8):1383-1397.
- 156. Funamoto K, Hayase T, Saijo Y, Yambe T. Numerical experiment of transient and steady characteristics of ultrasonic-measurementintegrated simulation in three-dimensional blood flow analysis. *Ann Biomed Eng.* 2009;37(1):34-49.
- 157. Bertoglio C. Forward and Inverse Problems in Fluid-Structure Interaction. Application to Hemodynamics. PhD thesis. Université Pierre et Marie Curie; 2012.
- 158. Funamoto K, Hayase T. Reproduction of pressure field in ultrasonic-measurement-integrated simulation of blood flow. *Int J Numer Methods Biomed Eng.* 2013;29(7):726-740.
- 159. Ii S, Adib MAHM, Watanabe Y, Wada S. Physically consistent data assimilation method based on feedback control for patient-specific blood flow analysis. *Int J Numer Methods Biomed Eng.* 2018;34(1):e2910.
- 160. Bertoglio C, Chapelle D, Fernández MA, Gerbeau JF, Moireau P. State observers of a vascular fluid-structure interaction model through measurements in the solid. *Comput Methods Appl Mech Eng.* 2013;256:149-168.
- 161. Moireau P, Chapelle D, Le Tallec P. Joint state and parameter estimation for distributed mechanical systems. *Comput Methods Appl Mech Eng.* 2008;197:659-677.
- Moireau P, Chapelle D, Le Tallec P. Filtering for distributed mechanical systems using position measurements: perspective in medical imaging. *Inverse Probl.* 2009;25(3):35010-35035. doi:10.1088/0266-5611/25/3/035010
- 163. Failer L. Optimal Control of Time-Dependent Nonlinear Fluid-Structure Interaction. PhD thesis. Technische Universität München; 2017.
- 164. Corrado C, Gerbeau JF, Moireau P. Identification of weakly coupled multiphysics problems. Application to the inverse problem of electrocardiography. *J Comput Phys.* 2015;283:271-298.
- 165. Pant S, Corsini C, Baker C, Hsia TY, Pennati G, Vignon-Clementel IE. Data assimilation and modelling of patient-specific singleventricle physiology with and without valve regurgitation. *J Biomech.* 2016;49(11):2162-2173. doi:10.1016/j.jbiomech.2015.11.030
- 166. Pant S, Corsini C, Baker C, Hsia TY, Pennati G, Vignon-Clementel IE. Inverse problems in reduced order models of cardiovascular Haemodynamics: aspects of data assimilation and heart rate variability. *J R Soc Interface*. 2017;14(126):20160513.

- 167. DeVault K, Gremaud PA, Novak V, Olufsen MS, Vernieres G, Zhao P. Blood flow in the circle of Willis: modeling and calibration. *Multiscale Model Simul.* 2008;7(2):888-909.
- 168. Lal R, Mohammadi B, Nicoud F. Data assimilation for identification of cardiovascular network characteristics. Int J Numer Methods Biomed Eng. 2017;33(5):e2824.
- 169. Caiazzo A, Caforio F, Montecinos G, Muller LO, Blanco PJ, Toro EF. Assessment of reduced-order unscented Kalman filter for parameter identification in 1-dimensional blood flow models using experimental data. *Int J Numer Methods Biomed Eng.* 2017;33(8):e2843.
- 170. Müller LO, Caiazzo A, Blanco PJ. Reduced-order unscented Kalman filter with observations in the frequency domain: application to computational hemodynamics. *IEEE Trans Biomed Eng.* 2018;66:1269-1276.
- 171. Lassila T, Manzoni A, Quarteroni A, Rozza G. A reduced computational and geometrical framework for inverse problems in hemodynamics. *Int J Numer Methods Biomed Eng.* 2013;29(7):741-776.
- 172. Arnold A, Battista C, Bia D, et al. Uncertainty quantification in a patient-specific one-dimensional arterial network model: EnKF-based inflow estimator. *J Verif, Valid Uncert Quantif.* 2017;2(1):011002.
- 173. Paun LM, Colebank MJ, Olufsen MS, Hill NA, Husmeier D. Assessing model mismatch and model selection in a Bayesian uncertainty quantification analysis of a fluid-dynamics model of pulmonary blood circulation. *J R Soc Interface*. 2020;17(173):20200886.
- 174. Paun LM, Husmeier D. Markov chain Monte Carlo with Gaussian processes for fast parameter estimation and uncertainty quantification in a 1D fluid-dynamics model of the pulmonary circulation. *Int J Numer Methods Biomed Eng.* 2021;37(2):e3421.
- 175. Schiavazzi DE, Arbia G, Baker C, et al. Uncertainty quantification in virtual surgery hemodynamics predictions for single ventricle palliation. Int J Numer Methods Biomed Eng. 2016;32(3):e02737. doi:10.1002/cnm.2737
- 176. Tran JS, Schiavazzi DE, Ramachandra AB, Kahn AM, Marsden AL. Automated tuning for parameter identification and uncertainty quantification in multi-scale coronary simulations. *Comput Fluids*. 2017;142:128-138.
- 177. Larson K, Bowman C, Papadimitriou C, Koumoutsakos P, Matzavinos A. Detection of arterial wall abnormalities via Bayesian model selection. *R Soc Open Sci.* 2019;6(10):182229. doi:10.1098/rsos.182229
- 178. Rispoli VC, Nielsen JF, Nayak KS, Carvalho JL. Computational fluid dynamics simulations of blood flow regularized by 3D phase contrast MRI. *Biomed Eng Online*. 2015;14(1):110.
- 179. Fathi MF, Bakhshinejad A, Baghaie A, et al. Denoising and spatial resolution enhancement of 4D flow MRI using proper orthogonal decomposition and lasso regularization. *Comput Med Imaging Graph*. 2018;70:165-172.
- Galarce F, Gerbeau JF, Lombardi D, Mula O. Fast reconstruction of 3D blood flows from Doppler ultrasound images and reduced models. *Comput Methods Appl Mech Eng.* 2021;375:113559.
- 181. Galarce F, Lombardi D, Mula O. Reconstructing haemodynamics quantities of interest from Doppler ultrasound imaging. *International journal for numerical methods. Biomed Eng.* 2020;37:e03416.
- 182. Galarce F, Lombardi D, Mula O. State estimation with model reduction and shape variability. Application to biomedical problems. arXiv:2106.09421; 2021.
- 183. Litjens G, Kooi T, Bejnordi BE, et al. A survey on deep learning in medical image analysis. Med Image Anal. 2017;42:60-88.
- 184. Zhou SK, Greenspan H, Shen D. Deep Learning for Medical Image Analysis. Academic Press; 2017.
- Hesamian MH, Jia W, He X, Kennedy P. Deep learning techniques for medical image segmentation: achievements and challenges. J Digit Imag. 2019;32(4):582-596.
- 186. Ker J, Wang L, Rao J, Lim T. Deep learning applications in medical image analysis. IEEE Access. 2017;6:9375-9389.
- 187. Shen D, Wu G, Suk HI. Deep learning in medical image analysis. Annu Rev Biomed Eng. 2017;19:221-248.
- Liang L, Mao W, Sun W. A feasibility study of deep learning for predicting hemodynamics of human thoracic aorta. J Biomech. 2020; 99:109544.
- Feiger B, Gounley J, Adler D, et al. Accelerating massively parallel hemodynamic models of Coarctation of the aorta using neural networks. Sci Rep. 2020;10(1):9508. doi:10.1038/s41598-020-66225-0
- 190. Yevtushenko P, Goubergrits L, Gundelwein L, et al. Deep learning based centerline-aggregated aortic hemodynamics: an efficient alternative to numerical modelling of hemodynamics. *IEEE J Biomed Health Inform*. 2021;26:1815-1825.
- 191. Raissi M, Perdikaris P, Karniadakis GE. Physics-informed neural networks: A deep learning framework for solving forward and inverse problems involving nonlinear partial differential equations. *J Comput Phys.* 2019;378:686-707. doi:10.1016/j.jcp.2018.10.045
- 192. Raissi M, Yazdani A, Karniadakis GE. Hidden fluid mechanics: learning velocity and pressure fields from flow visualizations. *Science*. 2020;367(6481):1026-1030.
- 193. Kissas G, Yang Y, Hwuang E, Witschey WR, Detre JA, Perdikaris P. Machine learning in cardiovascular flows modeling: predicting arterial blood pressure from non-invasive 4D flow MRI data using physics-informed neural networks. *Comput Methods Appl Mech Eng.* 2020;358:112623.
- 194. Fathi MF, Perez-Raya I, Baghaie A, et al. Super-resolution and denoising of 4D-flow MRI using physics-informed deep neural nets. *Comput Methods Programs Biomed.* 2020;197:105729.
- Regazzoni F, Chapelle D, Moireau P. Combining data assimilation and machine learning to build data-driven models for unknown long time dynamics—applications in cardiovascular modeling. *Int J Numer Methods Biomed Eng.* 2021;37(7):e3471. doi:10.1002/cnm. 3471
- 196. Romarowski RM, Lefieux A, Morganti S, Veneziani A, Auricchio F. Patient-specific CFD modelling in the thoracic aorta with PC-MRIbased boundary conditions: A least-square three-element Windkessel approach. *Int J Numer Methods Biomed Eng* 2018; 34(11):e3134. doi:10.1002/cnm.3134

- 197. Clark JW, Ling RYS, Srinivasan R, Cole JS, Pruett RC. A two-stage identification scheme for the determination of the parameters of a model of left heart and systemic circulation. *IEEE Trans Biomed Eng*. 1980;27(1):20-29. doi:10.1109/TBME.1980.326687
- 198. Burattini R, Gnudi G. Computer identification of models for the arterial tree input impedance: comparison between two new simple models and first experimental results. *Med Biol Eng Comput.* 1982;20(2):134-144.
- Toorop GP, Westerhof N, Elzinga G. Beat-to-beat estimation of peripheral resistance and arterial compliance during pressure transients. Amer J Physiol-Heart Circulat Physiol. 1987;252(6):H1275-H1283.
- 200. Segers P, Rietzschel ER, De Buyzere ML, et al. Three- and four-element Windkessel models: assessment of their fitting performance in a large cohort of healthy middle-aged individuals. *Proceed Instit Mech Eng, H: J Eng Med.* 2008;222(4):417-428. doi:10.1243/ 09544119JEIM287
- Hann CE, Chase JG, Desaive T, et al. Unique parameter identification for cardiac diagnosis in critical care using minimal data sets. Comput Methods Programs Biomed. 2010;99(1):75-87. doi:10.1016/j.cmpb.2010.01.002
- Revie JA, Stevenson DJ, Chase JG, et al. Validation of subject-specific cardiovascular system models from porcine measurements. Comput Methods Programs Biomed. 2013;109(2):197-210. doi:10.1016/j.cmpb.2011.10.013
- 203. Sughimoto K, Liang F, Takahara Y, et al. Assessment of cardiovascular function by combining clinical data with a computational model of the cardiovascular system. J Thorac Cardiovasc Surg. 2013;145(5):1367-1372. doi:10.1016/j.jtcvs.2012.07.029
- Pironet A, Desaive T, Dauby PC, Chase JG, Docherty PD. Parameter identification methods in a model of the cardiovascular system. IFAC-PapersOnLine. 2015;48(20):366-371.
- Schiavazzi DE, Baretta A, Pennati G, Hsia TY, Marsden AL. Patient-specific parameter estimation in single-ventricle lumped circulation models under uncertainty. Int J Numer Methods Biomed Eng. 2016;33:e02799. doi:10.1002/cnm.2799
- 206. Cousins W, Gremaud PA. Impedance boundary conditions for general transient hemodynamics. Int J Numer Methods Biomed Eng. 2014;30(11):1294-1313. doi:10.1002/cnm.2658
- 207. Olufsen MS. Structured tree outflow condition for blood flow in larger systemic arteries. *Amer J Physiol–Heart Circulat Physiol*. 1999; 276(1):H257-H268.
- Blanco PJ, Watanabe SM, Feijóo RA. Identification of vascular territory resistances in one-dimensional hemodynamics simulations. J Biomech. 2012;45(12):2066-2073. doi:10.1016/j.jbiomech.2012.06.002
- Ventre J, Politi MT, Fernández JM, et al. Parameter estimation to study the immediate impact of aortic cross-clamping using reduced order models. Int J Numer Methods Biomed Eng. 2021;37(11):e3261. doi:10.1002/cnm.3261
- 210. Carson J, Warrander L, Johnstone E, van Loon R. Personalising cardiovascular network models in pregnancy: A two-tiered parameter estimation approach. *International Journal for Numerical Methods in Biomedical Engineering*. 2021;37(11):e3267. doi:10.1002/cnm.3267
- 211. Quarteroni A, Rozza G. Optimal control and shape optimization of aorto-coronaric bypass anastomoses. *Math Models Methods Appl Sci.* 2003;13(12):1801-1823.
- Agoshkov V, Quarteroni A, Rozza G. A mathematical approach in the design of arterial bypass using unsteady stokes equations. J Scient Comput. 2006;28(2–3):139-165.
- Manzoni A, Ponti L. An adjoint-based method for the numerical approximation of shape optimization problems in presence of fluidstructure interaction. ESAIM. Math Model Numer Anal. 2018;52(4):1501-1532.
- Aguayo J, Bertoglio C, Osses A. A distributed resistance inverse method for flow obstacle identification from internal velocity measurements. *Inverse Probl.* 2021;37(2):025010.

How to cite this article: Nolte D, Bertoglio C. Inverse problems in blood flow modeling: A review. *Int J Numer Meth Biomed Engng*. 2022;38(8):e3613. doi:10.1002/cnm.3613